PARTNERS REAL ESTATE INVESTMENT TRUST
AMENDED AND RESTATED DECLARATION OF TRUST
March 27, 2007, as amended and restated with effect as of June 4, 2010 and as further amended and restated with effect as of November 3, 2010 and as further amended and restated with effect as of December 15, 2011 and as further amended and restated with effect as of March 29, 2012

TABLE OF CONTENTS

ARTIC	LE 1 INTERPRETATION	
1.1	DEFINITIONS	
1.2	REFERENCES TO ACTS PERFORMED BY THE TRUST	7
1.3	References	8
1.4	Number and Gender	8
1.5	HEADINGS FOR REFERENCE ONLY	8
1.6	DAY NOT A BUSINESS DAY	8
1.7	CERTAIN PHRASES, ETC.	
1.8	GOVERNING LAW	
1.9	CURRENCY	9
1.10		
1.11	TAX ACT	
ARTIC	CLE 2 DECLARATION OF TRUST	9
2.1	ESTABLISHMENT OF THE TRUST	9
2.2	INITIAL CONTRIBUTION	
2.3	NAME OF THE TRUST	9
2.4	HEAD OFFICE	10
2.5	NATURE OF THE TRUST	
2.6	RIGHTS OF UNITHOLDERS AND SPECIAL VOTING UNITHOLDERS	10
2.7	UNITHOLDERS AND SPECIAL VOTING UNITHOLDERS BOUND	
2.8	LIABILITY OF UNITHOLDERS	11
ARTIC	CLE 3 ISSUE AND SALE OF UNITS	
3.1	NATURE OF UNITS AND SPECIAL VOTING UNITS	
3.2	AUTHORIZED NUMBER OF UNITS AND SPECIAL VOTING UNITS	
3.3	ISSUE OF UNITS	13
3.4	FRACTIONAL UNITS	14
3.5	RE-PURCHASE OF UNITS BY TRUST	14
3.6	CONSOLIDATION OF UNITS	15
3.7	Pre-Emptive Rights	15
3.8	UNCLAIMED DISTRIBUTIONS	15
ARTIC	LE 4 INVESTMENTS AND OPERATIONS OF THE TRUST	
4.1	INVESTMENT GUIDELINES	
4.2	OPERATING POLICIES	18
4.3	TAX STATUS	19
4.4	APPLICATION OF INVESTMENT GUIDELINES AND OPERATING GUIDELINES	
4.5	REGULATORY MATTERS	20
ARTIC	CLE 5 DISTRIBUTIONS	
5.1	COMPUTATION OF INCOME AND NET REALIZED CAPITAL GAINS	
5.2	DISTRIBUTIONS	
5.3	OTHER DISTRIBUTIONS	
5.4	CHARACTER OF DISTRIBUTIONS AND DESIGNATIONS	
5.5	ENFORCEABILITY OF RIGHT TO RECEIVE DISTRIBUTIONS	
5.6	METHOD OF PAYMENT OF DISTRIBUTIONS	23
5.7	WITHHOLDING TAXES.	23
5.8	DEFINITIONS	23
5.9	PAYMENTS OF CASH	
5.10	DISTRIBUTION REINVESTMENT AND UNIT PURCHASE PLAN	24
5.11	INCOME TAX MATTERS	24
ARTIC	CLE 6 REDEMPTION OF UNITS	24

6.1	REDEMPTION RIGHT	24
	EXERCISE OF REDEMPTION RIGHT	
	Cash Redemption	
	NO CASH REDEMPTION IN CERTAIN CIRCUMSTANCES	
	N SPECIE REDEMPTION	
	CANCELLATION OF ALL REDEEMED UNITS	
6.7	WITHHOLDINGS BY THE TRUSTEES	27
6.8	REDEMPTION OF INITIAL UNIT	27
ARTICL	E 7 REDEMPTION OF SPECIAL VOTING UNITS	27
	RIGHT OF REDEMPTION	
	EXERCISE OF REDEMPTION RIGHT	
	Cash Redemption	
	CANCELLATION OF ALL REDEEMED SPECIAL VOTING UNITS	
ARTICI	E 8 TRUSTEES	28
	NUMBER OF TRUSTEES	
	CALLING AND NOTICE OF MEETINGS	
	PLACE OF MEETINGS.	
	MEETINGS BY TELEPHONE	
	QUORUM	
	CHAIR	
	ACTION BY THE TRUSTEES	
	ADJOURNED MEETING	
	REMUNERATION AND EXPENSES.	
8.10	OFFICERS	
8.11	COMMITTEES	
8.12	COMMITTEES COMMITTEE PROCEDURE.	
	E 9 APPOINTMENT, RESIGNATION AND REMOVAL OF THE TRUSTEES	
	QUALIFICATION OF TRUSTEES	
	PROPOSED MANAGER TRUSTEES	
	ELECTION OF TRUSTEES	
	CONSENT TO ACT	
	FAILURE TO ELECT MINIMUM NUMBER OF TRUSTEES	
	CEASING TO HOLD OFFICE	
	FILLING VACANCIES	
7.7	VALIDITY OF ACTS	
	E 10 CONCERNING THE TRUSTEES	
10.1	POWERS OF THE TRUSTEES	
10.2	SPECIFIC POWERS AND AUTHORITIES	
10.3	FURTHER POWERS OF THE TRUSTEES	
10.4	BANKING	
10.5	STANDARD OF CARE AND DUTIES	
10.6	FEES AND EXPENSES	
10.7	DETERMINATIONS OF TRUSTEES BINDING	
10.8	LIMITATIONS ON LIABILITY	
10.9	INDEMNIFICATION	
10.10	CONTRACTUAL OBLIGATIONS OF THE TRUST	
10.11	CONFLICTS OF INTEREST.	
10.12	CONDITIONS PRECEDENT	
10.13 10.14	EXECUTION OF INSTRUMENTS AND APPARENT AUTHORITY DECLARATIONS AS TO BENEFICIAL OWNER	
	E 11 AMENDMENT	16

11.1	AMENDMENT	
11.2	SPECIAL RESOLUTION	
11.3	Variation of Rights	
11.4	Internal Restructuring	
11.5	No Termination	49
ARTICL	E 12 MEETINGS OF VOTING UNITHOLDERS	
12.1	ANNUAL AND SPECIAL MEETINGS OF VOTING UNITHOLDERS	
12.2	NOTICE OF MEETINGS	51
12.3	Quorum	
12.4	Voting Rights	
12.5	RESOLUTIONS BINDING THE TRUSTEES	
12.6	MEANING OF "SPECIAL RESOLUTION"	
12.7	Meaning of "Outstanding"	
12.8	RECORD DATE FOR VOTING	
12.9	Proxies	
12.10	Personal Representatives	
12.11	CONDUCT OF MEETINGS	
12.12	BINDING EFFECT OF RESOLUTIONS	
12.13	RESOLUTION IN WRITING	55
ARTICL	E 13 CERTIFICATES AND REGISTRATION OF UNITS AND SPECIAL VOTING UNITS	
13.1	NATURE AND EVIDENCE OF VOTING UNITS	56
13.2	Liability	
13.3	Unit Certificates	
13.4	CONTENTS OF UNIT CERTIFICATES	
13.5	REGISTER OF UNITHOLDERS AND SPECIAL VOTING UNITHOLDERS	
13.6	Transfer Agents and Registrars	
13.7	LOST CERTIFICATES	59
	E 14 OWNERSHIP AND TRANSFER OF UNITS AND SPECIAL VOTING UNITS	
14.1	LIMITATION OF NON-RESIDENT OWNERSHIP	
14.2	Transfer of Units	
14.3	Units and Special Voting Units Held Jointly or in a Fiduciary Capacity	
14.4	Performance of Trust	
14.5 14.6	DEATH OR DISABILITY OF A UNITHOLDER OR SPECIAL VOTING UNITHOLDEROFFER FOR UNITS	
14.0	OFFER FOR UNITS	03
	E 15 TERMINATION	
15.1	TERM OF TRUSTEXTENSION OR TERMINATION WITH THE APPROVAL OF VOTING UNITHOLDERS	
15.2	PROCEDURE UPON TERMINATION	
15.3 15.4	PROCEDURE UPON TERMINATION POWERS OF THE TRUSTEES UPON TERMINATION	
15.4	SALE OF INVESTMENTS	
15.5	DISTRIBUTION OF PROCEEDS OR ASSETS	
15.7	FURTHER NOTICE TO VOTING UNITHOLDERS	
15.8	RESPONSIBILITY OF THE TRUSTEES AFTER SALE AND CONVERSION	
ADTICI	E 16 SUPPLEMENTAL INDENTURES	۷0
16.1	PROVISION FOR SUPPLEMENTAL INDENTURES FOR CERTAIN PURPOSES	
ADTICI	E 17 GENERAL	<i>(</i> 0
17.1	NOTICES	
17.1	FAILURE TO GIVE NOTICE	
17.2	JOINT HOLDERS	
17.3 17.4	SERVICE OF NOTICE	69 60

17.5	INFORMATION AVAILABLE TO VOTING UNITHOLDERS	70
17.6	FISCAL YEAR	70
17.7	Financial Disclosure	70
17.8	VOTING UNITHOLDER MEETING INFORMATION	70
17.9	TAXATION INFORMATION	71
17.10	DELEGATION OF TRUST MANAGEMENT	71
17.11	Power of Attorney	71
17.12	No Breach	72
17.13	EXECUTION OF INSTRUMENTS	72
17.14	Trust Property	72
17.15	Trust Records	72
ARTICL	Æ 18 AUDITORS	73
ARTICL 18.1	E 18 AUDITORS	
		73
18.1	QUALIFICATION OF AUDITORS	
18.1 18.2	QUALIFICATION OF AUDITORS	
18.1 18.2 18.3	QUALIFICATION OF AUDITORS APPOINTMENT OF AUDITORS CHANGE OF AUDITORS	
18.1 18.2 18.3 18.4	QUALIFICATION OF AUDITORS APPOINTMENT OF AUDITORS CHANGE OF AUDITORS	
18.1 18.2 18.3 18.4	QUALIFICATION OF AUDITORS APPOINTMENT OF AUDITORS CHANGE OF AUDITORS REPORT OF AUDITORS	
18.1 18.2 18.3 18.4 ARTICL	QUALIFICATION OF AUDITORS APPOINTMENT OF AUDITORS CHANGE OF AUDITORS REPORT OF AUDITORS E 19 MISCELLANEOUS	
18.1 18.2 18.3 18.4 ARTICL 19.1	QUALIFICATION OF AUDITORS APPOINTMENT OF AUDITORS CHANGE OF AUDITORS REPORT OF AUDITORS .E 19 MISCELLANEOUS COUNTERPARTS	73 73 73 73 73 73 73
18.1 18.2 18.3 18.4 ARTICL 19.1 19.2	QUALIFICATION OF AUDITORS APPOINTMENT OF AUDITORS CHANGE OF AUDITORS REPORT OF AUDITORS E 19 MISCELLANEOUS COUNTERPARTS SEVERABILITY	73 73 73 73 73 73 74

PARTNERS REAL ESTATE INVESTMENT TRUST

AMENDED AND RESTATED DECLARATION OF TRUST

THIS DECLARATION OF TRUST made in Toronto, Ontario the 27th day of March, 2007 and amended and restated in Toronto, Ontario with effect as of the 4th day of June, 2010 and further amended and restated with effect as of the 3rd day of November, 2010 and further amended and restated with effect as of the 15th day of December, 2011 and further amended and restated with effect as of March 29, 2012.

BETWEEN:

Paul Dykeman, resident in the Province of Nova Scotia

Adam Gant, resident in Province of British Columbia

John van Haastrecht, resident in the Province of Ontario

Patrick Miniutti, resident in the Province of British Columbia

Saul Shulman, resident in the Province of Ontario and

Louis Maroun, resident in Bermuda,

all of whom are the trustees of the trust as further amended and restated herein (collectively, the "2010 Trustees"), and each person who after the date hereof becomes a trustee of the trust as herein provided,

- and -

all persons who after the date hereof become holders of units of the Trust as herein provided,

WHEREAS Charter Real Estate Investment Trust ("Charter REIT") was settled on March 27, 2007 with \$10.00 (the "Initial Contribution") by the C.A. Bancorp Inc., (the "Initial Unitholder") in consideration for the issuance by the initial Trustees to the Initial Unitholder of one Unit;

AND WHEREAS the Trustees have resolved to change the name of Charter REIT to "Partners Real Estate Investment Trust" (the "**Trust**") pursuant to section 2.3 of amended and restated declaration of trust;

AND WHEREAS the Trust has been established for the principal purpose of providing persons who may become the holders of Units of the Trust with an opportunity to participate directly or indirectly in a portfolio of income-producing real property investments and related assets;

AND WHEREAS the 2010 Trustees replaced the initial Trustees as authorized pursuant to this Declaration of Trust;

AND WHEREAS the initial Trustees originally agreed and the 2010 Trustees have since agreed in their place to hold the Initial Contribution and all amounts and assets subsequently received under this Declaration of Trust or in respect of the investment of the assets of the Trust in accordance with the provisions hereinafter set forth;

AND WHEREAS the 2010 Trustees desire and the Initial Unitholder similarly desired that the Trust shall qualify as a "mutual fund trust" pursuant to subsection 132(6) of the Tax Act;

AND WHEREAS pursuant to Article 11 of the Declaration of Trust, the 2010 Trustees are authorized to make amendments to the Declaration of Trust, as provided for therein;

AND WHEREAS the 2010 Trustees wish to amend and restate the Declaration of Trust as herein provided, which such amendments have been approved by Special Resolution of the Unitholders, and which set out the agreements, terms and conditions which shall govern their mutual and respective rights, powers and obligations with respect to the settlement and administration of the Trust.

NOW THEREFORE, each of the undersigned 2010 Trustees, hereby confirms and declares that he holds in trust the Initial Contribution and any and all other property, real, personal or otherwise, tangible or intangible, which has been at the date hereof or is hereafter transferred, conveyed or paid to or otherwise received by him as Trustee or to which the Trust is otherwise entitled and all rents, income, profits and gains therefrom for the benefit of the Unitholders hereunder in accordance with and subject to the expressed provisions of this Declaration of Trust as amended and restated, to wit:

ARTICLE 1 INTERPRETATION

1.1 Definitions

In this Declaration of Trust, including the recitals hereto, unless the context otherwise requires, the following terms shall have the following meanings:

"Adjusted Unitholders' Equity" means, at any time, the aggregate of Unitholders' equity and the amount of accumulated depreciation and amortization recorded on the books and records of the Trust in respect of its properties and assets, calculated in accordance with GAAP.

"affiliate", when used to indicate a relationship with a person, has the meaning ascribed thereto in National Instrument 45-106 - *Prospectus and Registration Exemptions*.

"Arrangement" means the arrangement involving the Trust and Charter carried out pursuant to (i) an arrangement agreement among the same parties, as such agreement may be amended from time to time, (ii) a plan of arrangement under Section 193 of the *Business Corporations Act* (Alberta), and (iii) a final order of the Court of Queen's Bench in respect thereof.

"associate" has the meaning specified in the *Securities Act* (Ontario) as in effect on the date hereof.

- "Auditors" means the firm of chartered accountants appointed as the auditors of the Trust from time to time in accordance with the provisions hereof.
- "Beneficial Unitholder" means, in connection with the use of the Book-Entry System for the Units, the beneficial owner of a Unit.
- "Book-Entry System" means the record-entry securities transfer and pledge system known, as of the date hereof, by such name, which is administered by CDS in accordance with the operating rules and procedures of the Securities Settlement Service of CDS in force from time to time, or any successor system which CDS may offer from time to time.
- "Business Day" means any day other than a Saturday, a Sunday or a statutory holiday in the City of Toronto, Ontario.
- "CDS" means The Canadian Depository for Securities Limited and its successors.
- "CDS Participant" means a broker, dealer, bank, other financial institution or other person who, directly or indirectly, from time to time affects book-based transfers with CDS and pledges of securities deposited with CDS.
- "Charter" means Charter Realty Holdings Ltd.
- "Closing" means the completion of the Arrangement, upon the filing of articles of arrangement in respect thereof.
- "Closing Date" means the date on which Closing occurs.
- "Closing Market Price" has the meaning specified in Section 6.3(1).
- "Counsel" means a barrister or solicitor or firm of barristers and solicitors or other lawyers in an appropriate jurisdiction retained by the Trust, who may be counsel for the Trust or any of its Unitholders.
- "CPOA" has the meaning specified in Section 9.6(4).
- "**Damages**" has the meaning specified in Section 10.9(1).
- "Declaration of Trust" means this declaration of trust, as the same may be further supplemented, amended or restated from time to time; and "hereto", "herein", "hereof", "hereby", "hereunder" and similar expressions refer to this instrument and not to any particular Article, Section or portion hereof and the expressions "Article" and "Section" followed by a number mean and refer to the specified Article or Section of this Declaration of Trust.
- "Distribution Payment Date" means, in respect of a Distribution Period, a Business Day that is not later than 15 calendar days immediately following the end of a Distribution Period, or such other date as may be determined from time to time by the Trustees.

- "Distribution Period" means each calendar quarter or each calendar month, as determined by the Trustees from time to time, from and including the first day thereof and to and including the last day thereof; provided that (i) "Distribution Period" shall initially mean each calendar quarter and (ii) the first Distribution Period will begin on (and include) the Closing Date and will end on June 30, 2007.
- "Distribution Record Date" means, in respect of a Distribution Period, the last Business Day of the Distribution Period, or such other date as may be determined from time to time by the Trustees, except that December 31 shall in all cases be a Distribution Record Date.
- "**Distributions**" has the meaning specified in Section 5.2(1).
- "Exchange Agreement" means any exchange agreement entered into between the Trust and any other person with respect to the exchange of securities into Units.
- "Exchangeable Securities" means any securities that are exchangeable or convertible, directly or indirectly, for or into, as applicable, Units, that contain transfer restrictions acceptable to the Toronto Stock Exchange.
- "Fixed Number" has the meaning specified in Section 8.1.
- "GAAP" means at any time accounting principles generally accepted in Canada, including those set out in the Handbook of the Canadian Institute of Chartered Accountants, applied on a consistent basis.
- "Global Unit Certificate" has the meaning specified in Section 13.1(2).
- "Gross Book Value" means from time to time the book value of the assets of the Trust and its consolidated subsidiaries, as shown on its then most recent consolidated balance sheet, plus accumulated depreciation and amortization on assets and excluding the amount of any receivable reflecting interest rate subsidies on any debt assumed by the Trust, or, if approved by a majority of the Trustees, the most recently appraised value of the assets of the Trust and its consolidated subsidiaries may be used instead of book value.
- "Indemnified Person" has the meaning specified in Section 10.9(1).
- "Independent Trustee" means a Trustee who is "independent" (as defined in Multilateral Instrument 52-110 as in effect on the date hereof and as amended from time to time) to the Trust.
- "**Information Circular**" means the management information circular of Charter, with respect to the approval by its shareholders of the Arrangement and certain other matters.
- "Initial Contribution" has the meaning specified in the recitals to this Declaration of Trust.
- "**Initial Unitholder**" has the meaning specified in the first paragraph of this Declaration of Trust.
- "In specie Redemption Price" has the meaning specified in Section 6.5.

- "Management Agreement" means the management agreement dated as of June 4, 2010 and made among Charter Real Estate Investment Trust (now Partners Real Estate Investment Trust) and the Manager, as such agreement may be assigned to the Trust and amended, restated, superseded or replaced from time to time.
- "Manager" means the manager of the Trust as appointed by the Trustees from time to time in accordance with this Declaration of Trust and any duly appointed successor.
- "Market Price" has the meaning specified in Section 6.3(1).
- "Monthly Limit" has the meaning specified in Section 6.4(a).
- "mortgage" means any mortgage, charge, hypothec, bond, debenture, note or other evidence of indebtedness directly or indirectly secured by real property.
- "Net Realized Capital Gains" has the meaning specified in Section 5.1.
- "Non-resident" means a person who is a non-resident of Canada or a partnership that is not a Canadian partnership for purposes of the Tax Act.
- "Ordinary Resolution" means a resolution proposed to be passed as an ordinary resolution at a meeting of Voting Unitholders (including an adjourned meeting) duly convened for that purpose and held in accordance with the provisions hereof at which a quorum is present, which resolution is passed by the affirmative votes of a majority of the Voting Units represented at the meeting in person or by proxy and voted upon such resolution or a resolution proposed to be passed as an ordinary resolution approved in writing by a majority of the holders of Voting Units.
- "person" means a natural person, partnership, limited partnership, limited liability partnership, corporation, joint stock company, trust, unincorporated association, limited liability company, joint venture or other entity or governmental or regulatory authority or entity, and pronouns have a similarly extended meaning.
- "Proposed Manager Trustee" has the meaning specified in Section 9.2.
- "Proposed Manager Trustee Notice" has the meaning specified in Section 9.2.
- "real property" means property which in law is real property and includes, whether or not the same would in law be real property, leaseholds, mortgages, undivided joint interests in real property (whether by way of tenancy in common, joint tenancy, co-ownership, joint venture or otherwise), any interests in any of the foregoing and securities of corporations, trusts or partnerships or other persons whose sole or principal purpose and activity is to invest in, hold and/or deal in real property.
- "Redemption Notes" means unsecured subordinated promissory notes of the Trust having a maturity date to be determined at the time of issuance by the Trustees (provided that in no event shall the maturity date be set at a date subsequent to the first Business Day following the fifth anniversary of the date of issuance of such note), bearing interest from the date of issue at a market rate of interest determined at the time of issuance by the Trustees, payable

for each month during the term on the 15th day of each subsequent month with all principal being due on maturity, such promissory notes to provide that the Trust shall at any time be allowed to prepay all or any part of the outstanding principal without notice or bonus.

"Redemption Price" has the meaning specified in Section 6.3(1).

"Register" has the meaning specified in Section 13.5.

"Registrar and Transfer Agency Agreement" means the agreement to be entered into on Closing between the Trust and the Transfer Agent.

"Resident Canadian" means a person resident in Canada, or a partnership that is a Canadian partnership, for purposes of the Tax Act.

"SIFT Proposals" has the meaning specified in Section 1.11.

"SIFT Trust" has the meaning ascribed thereto in the SIFT Proposals or any final legislation implementing the SIFT Proposals.

"Special Resolution" has the meaning specified in Section 12.6.

"Special Voting Unitholder" means at any time a person whose name appears on the Register as a holder of one or more Special Voting Units.

"Special Voting Units" has the meaning specified in Section 3.1.

"subsidiary" when used to indicate a relationship with a person, has the meaning ascribed thereto in National Instrument 45-106 – *Prospectus and Registration Exemptions*.

"Tax Act" means the *Income Tax Act*, R.S.C. 1985 (5th Supp.), c. 1 and the regulations thereunder.

"Transfer Agent" means any registrar and transfer agent as may from time to time be appointed by the Trust to act as registrar and transfer agent of Units, which initially will be Computershare Investor Services Inc., together with any sub-transfer agent duly appointed by the Transfer Agent.

"Trust" means the trust established and constituted by this Declaration of Trust.

"**Trust Assets**" means the properties and assets held from time to time by the Trust or by the Trustees on behalf of the Trust, including:

- (a) the Initial Contribution;
- (b) all funds, securities or other property derived or received from the issuance or sale of Units or other funds, securities or other property received by the Trustees in their capacity as trustees of the Trust;

- (c) any securities, whether debt or equity, of Charter or any subsidiaries of the Trust held from time to time by or on behalf of the Trust;
- (d) any investments held from time to time by or on behalf of the Trust;
- (e) any proceeds of disposition of any of the foregoing property or in respect of the investment or substitution of the foregoing property; and
- (f) all proceeds, income, dividends, interest, profit, return of capital, gains and accretions and all substituted assets, rights and benefits of any kind or nature whatsoever arising directly or indirectly from or in connection with or accruing to such foregoing property or such proceeds of disposition.

"Trust Income" has the meaning specified in Section 5.1.

"Trust Liabilities" has the meaning specified in Section 2.8.

"**Trustee**" means, at any time, a person who is, in accordance with the provisions hereof, a trustee of the Trust at such time, and "**Trustees**" means, at any time, all of the persons, each of whom is at that time a Trustee.

"Trustees' Regulations" has the meaning specified in Section 10.3.

"Unit Certificate" means a certificate, in the form approved by the Trustees, evidencing one or more Units, issued and certified in accordance with the provisions hereof.

"Unitholder" means, at any time, a holder at that time of one or more Units, as shown on the Register.

"Unit Option Plan" means any Unit compensation plan established by the Trustees pursuant to which options to purchase Units may be granted to Trustees and to officers, employees and consultants of the Trust, its affiliates and the Manager;

"Units" has the meaning specified in Section 3.1.

"Voting Unitholders" means the Unitholders and the Special Voting Unitholders, collectively.

"Voting Units" means the Units and the Special Voting Units.

1.2 References to Acts Performed by the Trust

Where any reference is made in this Declaration of Trust to an act to be performed by the Trust, such reference shall be construed and applied for all purposes as if it referred to an act to be performed by the Trustees on behalf of the Trust or by some other person duly authorized to do so by the Trustees or pursuant to the provisions hereof, and where any reference is made in this Declaration of Trust to actions, rights or obligations of the Trustees, such reference shall be construed and applied for all purposes to refer to actions, rights or obligations of the Trustees in their

capacity as trustees of the Trust, and not in their other capacities, unless the context otherwise requires.

1.3 References

Except as otherwise specifically provided herein, reference in this Declaration of Trust to any contract, agreement or any other instrument will be deemed to include references to the same as varied, amended, supplemented, restated or replaced from time to time. Except as otherwise specifically provided herein, any reference in this Declaration of Trust to a statute or regulations, rules, policies or instruments thereunder will be deemed to be a reference to such statute or regulation, rule, policy or instrument as amended, re-enacted or replaced from time to time, and reference to specific parts, paragraphs or sections thereof will include all amendments, re-enactments or replacements.

1.4 Number and Gender

In this Declaration of Trust, unless herein otherwise expressly provided or unless the context otherwise requires, words importing the singular number include the plural, and vice versa; words importing a gender shall include the feminine, masculine and neuter genders; and words importing persons include an individual, partnership, association, body corporate, trustee, executor, administrator or legal representative.

1.5 Headings for Reference Only

The division of this Declaration of Trust into Sections, the provision of a Table of Contents and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Declaration of Trust.

1.6 Day Not a Business Day

If any day on which any amount is to be determined or any action is required to be taken hereunder is not a Business Day, then such amount will be determined or such action will be required to be taken at or before the requisite time, if any, on the next succeeding day that is a Business Day. This Section is not applicable to Section 5.1 and Section 5.2.

1.7 Certain Phrases, etc.

In this Declaration of Trust (i) the words "including" and "includes" mean "including (or includes) without limitation", and (ii) the phrase "the aggregate of", "the sum of", or a phrase of similar meaning means "the aggregate (or total or sum), without duplication, of".

1.8 Governing Law

This Declaration of Trust and the Unit Certificates shall be construed in accordance with the laws of the Province of Ontario and the laws of Canada applicable therein. Any and all disputes arising under this Declaration of Trust, whether as to interpretation, performance or otherwise, shall be subject to the jurisdiction of the courts of the Province of Ontario and each of the Trustees hereby irrevocably attorns, and each Unitholder shall be deemed to hereby irrevocably attorn, to the jurisdiction of the courts of such province.

1.9 Currency

Unless otherwise specified, all references to money amounts are to lawful currency of Canada.

1.10 Accounting Principles

All accounting terms not specifically defined in this Declaration of Trust will be interpreted in accordance with GAAP. Where the character or amount of any asset or liability or item of revenue or expense or amount of equity is required to be determined, or any consolidation or other accounting computation is required to be made, for the purpose of this Declaration of Trust, such determination or calculation will, to the extent applicable and except as otherwise specified herein or as otherwise determined by the Trustees, be made in accordance with GAAP.

1.11 Tax Act

Any reference herein to a particular provision of the Tax Act shall include a reference to that provision as it may be renumbered or amended from time to time. Where there are proposals for amendments to the Tax Act which have not been enacted into law or proclaimed into force on or before the date on which such proposals are to become effective, the Trustees may take such proposals into consideration and apply the provisions hereof as if such proposals had been enacted into law and proclaimed into force, including without limitation certain draft legislation in respect of income trusts and other flow-through entities released by the Department of Finance on December 21, 2006 and any successor proposals to such draft legislation (the "SIFT Proposals").

ARTICLE 2 DECLARATION OF TRUST

2.1 Establishment of the Trust

The Trustees hereby declare themselves and agree to act as trustees of the Trust and to hold and administer the Trust Assets in trust for the benefit of the Unitholders on and subject to the terms and conditions of this Declaration of Trust.

2.2 Initial Contribution

The Initial Unitholder pays concurrently with the execution of the Declaration of Trust, the Initial Contribution to the Trustees for the purpose of settling the Trust. The Trustees acknowledge receipt of the Initial Contribution and issue one Unit to the Initial Unitholder.

2.3 Name of the Trust

- (1) The Trust shall be known and designated as "**Partners Real Estate Investment Trust**" and, whenever practicable, lawful and convenient, the property of the Trust shall be held and the affairs of the Trust shall be conducted and transacted under that name.
- (2) If the Trustees determine that the use of such name is not practicable, legal or convenient, the Trust may use such other designation or may adopt such other name as the Trustees

deem appropriate, and the Trust may hold property and conduct and transact its affairs under such other designation or name.

(3) The Trustees may approve and use a version of any name or designation used by the Trust in any foreign language.

2.4 Head Office

The principal office of the Trust will be located at such place or places in Canada as the Trustees may from time to time designate. The Trust may have such other offices or places in Canada for the conduct of its affairs as the Trustees may from time to time determine as necessary or desirable.

2.5 Nature of the Trust

- (1) The Trust is an unincorporated open-ended limited purpose investment trust. The Trust is not, will not be deemed to be and will not be treated as, a partnership, limited partnership, society, syndicate, association, joint venture, company, corporation or joint stock company, nor will the Trustees or any individual Trustee or the Unitholders or Special Voting Unitholders or any of them or any person be, or be deemed to be, treated in any way whatsoever as liable or responsible hereunder as partners, joint venturers or as that of principal and agent or as members of a society, syndicate, association, partnership or limited partnership or shareholders of a corporation or other joint stock company.
- (2) The relationship of the Unitholders to the Trustees will be solely that of beneficiaries of the Trust, and the rights of Unitholders and Special Voting Unitholders will be limited to those expressly conferred upon them by this Declaration of Trust.

2.6 Rights of Unitholders and Special Voting Unitholders

The rights of each Unitholder (including the right, if any, to call for or receive a distribution or division of property, assets, monies, funds, income, dividends and capital gains held, received or realized by the Trust) are limited to those contained herein and, except as provided herein, no Unitholder will be entitled to call for any partition or division of the Trust Assets or for a distribution of any particular asset forming part of the Trust Assets or of any particular monies or funds received by the Trust. The legal ownership of the Trust Assets and the right to conduct the activities of the Trust are vested exclusively in the Trustees, and no Unitholder or Special Voting Unitholder has or is deemed to have any right of ownership in any of the Trust Assets, except as specifically provided herein. Except as specifically provided herein, no Unitholder or Special Voting Unitholder will be entitled to interfere with or give any direction to the Trustees with respect to the affairs of the Trust or in connection with the exercise of any powers or authorities conferred upon the Trustees under this Declaration of Trust. The Units and the Special Voting Units will be personal property and will confer upon the holders thereof only the interest and rights specifically set forth in this Declaration of Trust.

2.7 Unitholders and Special Voting Unitholders Bound

This Declaration of Trust shall be binding upon all persons who become Unitholders or Special Voting Unitholders from time to time. By acceptance of a certificate representing any Units

or Special Voting Units or, during use of the Book-Entry System for the Units, upon completion of a purchase of a Unit or a Special Voting Unit, the holder thereof shall be deemed to agree to be bound, and shall be so bound, by this Declaration of Trust.

2.8 Liability of Unitholders

- (1) No Unitholder, Beneficial Unitholder or Special Voting Unitholder, in its capacity as such, shall incur or be subject to any personal liability, direct or indirect, absolute or contingent, in contract or in tort or of any other kind to any person and no resort shall be had to, nor shall recourse or satisfaction be sought from the private property of any Unitholder, Beneficial Unitholder or Special Voting Unitholder, in its capacity as such, for any liability whatsoever in connection with:
 - (a) the Trust Assets or the ownership, use, operation, acquisition or disposition thereof or exercise or enjoyment of the rights, privileges, conditions or benefits attached thereto, associated therewith or derived therefrom;
 - (b) the obligations or the activities or affairs of the Trust;
 - (c) any actual or alleged act or omission of the Trustees or by any other person in respect of the activities or affairs of the Trust (whether or not authorized by or pursuant to this Declaration of Trust);
 - (d) any act or omission of the Trustees or by any other person in the performance or exercise, or purported or attempted performance or exercise, of any obligation, power, discretion or authority conferred upon the Trustees or such other person in respect of the activities or affairs of the Trust (whether or not authorized by or pursuant to this Declaration of Trust);
 - (e) any transaction entered into by the Trustees or by any other person in respect of the activities or affairs of the Trust (whether or not authorized by or pursuant to this Declaration of Trust); or
 - (f) any taxes, levies, imposts or charges or fines, penalties or interest in respect thereof payable by the Trust or by the Trustees or by any other person (except a Unitholder or a Special Voting Unitholder to the extent required by applicable tax laws by reason of or arising out of his ownership of Units or Special Voting Units) as applicable on behalf of or in connection with the activities or affairs of the Trust

(collectively "**Trust Liabilities**") but rather the assets of the Trust are intended to be subject to levy or execution for satisfaction of such Trust Liabilities.

- (2) No Unitholder, Beneficial Unitholder or Special Voting Unitholder in its capacity as such shall be liable to indemnify the Trustees or any other person with respect to any Trust Liabilities.
- (3) To the extent that, notwithstanding the provisions of this Section 2.8 any Unitholder, Beneficial Unitholder or Special Voting Unitholder, in its capacity as such, may be determined by a judgment of a court of competent jurisdiction to be subject to or liable in respect of any Trust Liabilities, such Unitholder, Beneficial Unitholder or Special Voting Unitholder is entitled to

indemnity and reimbursement out of the Trust Assets to the full extent of such liability and for all costs of any litigation or other proceedings in which such liability has been determined, including, without limitation, all fees and disbursements of counsel. The rights accruing to a Unitholder, Beneficial Unitholder or Special Voting Unitholder under this Section 2.8(3) do not exclude any other rights to which such Unitholder, Beneficial Unitholder or Special Voting Unitholder may be lawfully entitled, nor does anything herein contained restrict the right of the Trustees to indemnify or reimburse a Unitholder, Beneficial Unitholder or Special Voting Unitholder out of the Trust Assets in any appropriate situation not specially provided herein but, for greater certainty, the Trustees have no liability to reimburse Unitholders, Beneficial Unitholders or Special Voting Unitholder for taxes assessed against them by reason of or arising out of their ownership of Units or Special Voting Units (as applicable).

ARTICLE 3 ISSUE AND SALE OF UNITS

3.1 Nature of Units and Special Voting Units

- (1) The beneficial interests in the Trust will be divided into interests of two classes, described and designated as "**Units**" and "**Special Voting Units**", respectively, which will be entitled to the rights and subject to the limitations, restrictions and conditions set out herein.
- (2) Each Unit represents an equal undivided beneficial interest in the Trust, in any distribution from the Trust (whether of Trust Income, Net Realized Capital Gains or other amounts) and in any net assets of the Trust in the event of the termination or winding-up of the Trust. All Units will rank among themselves equally and rateably without discrimination, preference or priority. Each Unit shall entitle the holder of record thereof to one vote at all meetings of Voting Unitholders or in respect of any written resolution of Voting Unitholders.
- No Special Voting Unit will be entitled to any interest or share in the Trust, in any distribution from the Trust (whether of Trust Income, Net Realized Capital Gains or other amounts) or in any net assets of the Trust in the event of the termination or winding-up of the Trust, other than the right to receive, upon the redemption of a Special Voting Unit, the amount of \$0.00001 for each Special Voting Unit so redeemed. Special Voting Units may be issued in series and will only be issued in connection with or in relation to, if the Trustees so determine, Exchangeable Securities. A Special Voting Unit will be issued together with, and will be attached to, if the Trustees so determine, each Exchangeable Security issued, and will be evidenced only by the certificates representing such Exchangeable Securities. Special Voting Units will not be transferable separately from the Exchangeable Securities to which they are attached. Each Special Voting Unit will be redeemable by the holder of the Special Voting Unit in accordance with Article 7. Each Special Voting Unit will entitle the holder of record thereof from time to time, as shown on the Register, to that number of votes at all meetings of Voting Unitholders that is equal to the number of Units into which the Exchangeable Security to which such Special Voting Unit is attached is, directly or indirectly, exchangeable or convertible. Upon the exchange, redemption or conversion of an Exchangeable Security for Units, the Special Voting Unit that is attached to such Exchangeable Security will immediately be cancelled without any further action of the Trustees, and the former holder of such Special Voting Unit will cease to have any rights with respect thereto.

- (4) Concurrently with the issuance of any Exchangeable Securities (and any attached Special Voting Units), the Trust will enter into such agreements, including Exchange Agreements, as may be necessary or desirable to properly give effect to the terms of the Exchangeable Securities, including, without limitation, to provide for the voting of all attached Special Voting Units and the automatic conversion, exercise, redemption or exchange of such Exchangeable Securities in the event of an Offer (as defined in Section 14.6) for the Units as provided in Section 14.6.
- (5) Issued and outstanding Units and Special Voting Units may be subdivided or consolidated from time to time by the Trustees without the approval of Voting Unitholders. In the event of a subdivision or consolidation of the Units or Special Voting Units, such consolidation shall not result in a disposition, redemption, cancellation, novation, or transfer of the relevant units and each holder shall remain a holder of the relevant units such that there is no change in the holder's interests, rights or privileges in the capital structure of the Trust.

3.2 Authorized Number of Units and Special Voting Units

- (1) The number of Units that are authorized and that may be issued hereunder is unlimited.
- (2) The number of Special Voting Units that are authorized and that may be issued hereunder is unlimited.

3.3 Issue of Units

- (1) Units and Special Voting Units may be allotted and issued by the Trust at such times, to such persons, for such consideration and on such terms and conditions as the Trustees may determine; provided, however, that Special Voting Units may only be issued in connection with or in relation to Exchangeable Securities. The Trustees may authorize the Trust to pay a reasonable commission to any person in consideration of such person purchasing or agreeing to purchase, whether absolutely or conditionally, Units from the Trust or from any other person or procuring or agreeing to procure purchasers, whether absolute or conditional, for Units.
- (2) The Trustees may create and issue rights, warrants (including so-called "special warrants" or "subscription receipts" which may be exercisable for no additional consideration), convertible or exchangeable securities (including Units issuable upon the exchange of securities of other issuers, including Exchangeable Securities) or options (including all types of share compensation or incentive programs) to subscribe for Units, which rights, warrants, convertible or exchangeable securities or options may be exercisable at such subscription price or prices and at such time or times and on such terms or conditions as the Trustees may determine. The rights, warrants, convertible or exchangeable securities or options so created may be issued for such consideration, or for no consideration, all as the Trustees may determine. A right, warrant, convertible or exchangeable security or option will not be a Unit and the holder thereof will not be a Unitholder.
- (3) Units are only to be issued as fully paid in money, property (including, without limitation, an obligation to pay consideration in installments) or services, and are not to be subject to future calls or assessments, except that Units to be issued under an offering following the Closing

may be issued for consideration payable in installments, and the Trust may take a security interest over such Units or other property for unpaid installments.

- (4) For greater certainty, Units may be issued by the Trust:
- (a) as contemplated by this Section 3.3,
- (b) pursuant to Section 5.6,
- (c) as contemplated by an Exchange Agreement,
- (d) pursuant to the Unit Option Plan, any Unitholder rights plans, distribution reinvestment plans and Unit purchase plans, incentive option plans or other compensation plans, if any, established by the Trust or any of its affiliates,
- (e) in order to acquire and as consideration for securities, including securities of or issued by Charter and subsidiaries of the Trust,
- (f) upon the exercise of any rights, warrants, convertible or exchangeable securities or options to acquire Units issued by the Trust or other issuers, including upon the exercise of any rights of exchange or conversion set out in any Exchangeable Securities,
- (g) pursuant to Section 3.6,
- (h) in satisfaction of any indebtedness of or borrowing by the Trust, and
- (i) making non-cash distributions to Unitholders, including pursuant to distribution reinvestment plans, if any, established by the Trust.

3.4 Fractional Units

Fractions of Units may be issued, including pursuant to distributions of additional Units to all Unitholders pursuant to Section 5.6. No certificates will be issued for fractional Units. Fractional Units shall not, except to the extent that they may represent in the aggregate one or more whole Units, entitle the holder thereof to notice of, or to attend or to vote at meetings of Unitholders. Subject to the foregoing, fractions of Units will carry the rights and be subject to the provisions hereof applicable to whole Units in the proportion that they bear to one Unit.

3.5 Re-Purchase of Units by Trust

The Trust may, from time to time, purchase Units (or other securities of the Trust which may be issued and outstanding from time to time) for cancellation in accordance with the requirements of applicable securities legislation and the rules prescribed from time to time under applicable stock exchange or regulatory policies. In connection with or immediately after the Closing, the Trust will purchase the initial Unit from the Initial Unitholder, and the Initial Unitholder will sell the initial Unit to the Trust, for a purchase price of \$10.00 and, upon the completion of such purchase and sale, the initial Unit will be cancelled and will no longer be outstanding for any of the purposes of this Declaration of Trust.

3.6 Consolidation of Units

Unless the Trustees determine otherwise, and subject to all necessary regulatory approvals, immediately after any pro rata distribution of additional Units to all Unitholders pursuant to Section 5.6, the number of outstanding Units will be consolidated such that each Unitholder will hold after such consolidation the same number of Units as the Unitholder held immediately prior to the distribution of such additional Units. In such case, each Unit Certificate representing a number of Units prior to the distribution of additional Units will be deemed to represent the same number of Units after the distribution of additional Units and the consolidation. Notwithstanding the foregoing, where tax is required to be withheld in respect of a Unitholder's share of a distribution, the Trust shall withhold from the cash portion of such distribution, if any, or the Unitholder shall make a cash payment to the Trust, of an amount equal to the amount of tax required to be remitted to the appropriate taxation authority by the Trust, or, if such withholding cannot be made by the Trust or such payment is not made by the Unitholder the consolidation will result in such Unitholder holding that number of Units equal to the product of (i) the sum of the number of Units held by such Unitholder prior to the distribution and the number of Units received by such Unitholder in connection with the distribution (net of Units withheld by the Trust to satisfy the Trust's withholding obligations) and (ii) a fraction, the numerator of which is the aggregate number of Units outstanding prior to the distribution and the denominator of which is the aggregate number of Units that would be outstanding following the distribution and before the consolidation if no withholding were required in respect of any part of the distribution payable to any Unitholders. Any Units so withheld shall either be delivered to the appropriate taxation authority or sold, in which case the net proceeds shall be remitted to the appropriate taxation authority. Such Unitholder will be required to surrender the Unit Certificates, if any, representing such Unitholder's original Units in exchange for a Unit Certificate representing such Unitholder's post-consolidation Units.

3.7 Pre-Emptive Rights

[intentionally deleted]

3.8 Unclaimed Distributions

In the event that the Trustees hold amounts to be paid to Unitholders under Article 5, Article 6 or Article 15 or otherwise which are unclaimed or which can not be paid for any reason, neither the Trustees nor any distribution disbursing agent shall be under any obligation to invest or reinvest the same and shall only be obligated to hold the same in a current or other non-interest bearing account with a chartered bank or trust company, pending payment to the person or persons entitled thereto. The Trustees shall, as and when required by law, and may any time prior to such required time, pay all or part of the distributions so held to a court in the province in which the Trust has its principal office or to the Public Guardian and Trustee of Ontario (or other similar government official or agency in the province in which the Trust has its principal office) whose receipt shall be a good and sufficient discharge of the obligations of the Trustees.

ARTICLE 4 INVESTMENTS AND OPERATIONS OF THE TRUST

4.1 Investment Guidelines

After the Closing, the assets of the Trust may be invested, directly or indirectly, only in accordance with the following investment guidelines:

- (a) subject to the remainder of this Section 4.1, the Trust may only invest, directly or indirectly, in:
 - (i) interests (including fee ownership and leasehold interest) in incomeproducing real property;
 - (ii) corporations, trusts, partnerships or other persons which principally have interests (including the ownership of leasehold interests) in income-producing real property (or activities relating or ancillary thereto); and
 - (iii) such other activities, properties or assets as are consistent with the other investment guidelines of the Trust or as may be approved by Unitholders in accordance with the terms of this Declaration of Trust;
- (b) notwithstanding anything in Sections 4.1 and 4.2, the Trust shall not make any investment, take any action or omit to take any action that would result in Units not being units of a "mutual fund trust" within the meaning of the Tax Act;
- (c) the Trust may, directly or indirectly, invest in a joint venture arrangement for the purposes of owning interests or investments otherwise permitted to be held by the Trust; provided that such joint venture arrangement contains terms and conditions which, in the opinion of the Trustees, are commercially reasonable, including without limitation such terms and conditions relating to restrictions on transfer and the acquisition and sale of the Trust's and any joint venturer's interest in the joint venture arrangement, provisions to provide liquidity to the Trust, such as buy-sell mechanisms and provisions that limit the liability of the Trust to third parties. For purposes of this Section 4.1, a joint venture arrangement is an arrangement between the Trust and one or more other persons ("joint venturers") pursuant to which the Trust, directly or indirectly, conducts an undertaking for one or more of the purposes set out above and in respect of which the Trust may hold its interest jointly or in common or in another manner with others either directly or through the ownership of securities of a corporation or other entity (a "joint venture entity"), including without limitation a general partnership, limited partnership or limited liability company;
- (d) except for temporary investments held in cash, deposits with a Canadian chartered bank or trust company registered under the laws of Canada or of a province of Canada, short-term government debt securities, or receivables under installment receipt agreements or money market instruments of, or guaranteed by, a Schedule I Canadian bank maturing within one year from the date of issue or except as

permitted pursuant to Sections 4.1(a), 4.1(c), 4.1(f), 4.1(h) and 4.1(i), the Trust may not hold securities other than (i) securities of any issuer referred to in Section 4.1(a), (ii) securities of a joint venture entity, (iii) securities of an entity wholly-owned by the Trust, which has been formed and operated solely for the purpose of holding a particular real property or real properties; and (iv) securities of persons described in Section 4.1(f) and provided further that, notwithstanding anything contained in the Declaration of Trust to the contrary, the Trust may acquire securities of other real estate investment trusts;

- (e) the Trust shall not invest in rights to or interests in mineral or other natural resources, including oil or gas, except as incidental to an investment in real property;
- (f) notwithstanding the provisions of Section 4.1(d) or any other provision to the contrary, but subject to the provisions of Section 4.3, the Trust may invest in operating businesses which are ancillary to the Trust's ownership of real property or acquire interests in limited partnerships or corporations which may operate businesses related to the Trust's real estate investments;
- (g) the Trust shall not invest directly in raw land for development and ownership or for other development projects, except:
 - (i) for the purpose of renovating or expanding existing properties or facilities on adjacent properties; or
 - (ii) for the purpose of developing new properties which will be or are expected to be, upon completion, income producing, provided that the aggregate value of investments in raw land for such purpose will not, after giving effect to the proposed investment, exceed 10% of the Adjusted Unitholders' Equity;
- (h) the Trust shall invest in a mortgage or a mortgage bond (including a participating or convertible mortgage) only where:
 - (i) the real property which is security therefor is income-producing real property which otherwise meets the investment guidelines of the Trust contained in this Section 4.1; and
 - (ii) the aggregate value of the investments of the Trust in mortgages and mortgage bonds, after giving effect to the proposed investment, will not exceed 20% of the Adjusted Unitholders' Equity;
- (i) notwithstanding any of the provisions of Section 4.1(h) hereof, the Trust may invest in any mortgage which is not a first ranking mortgage, including mezzanine financings, for purposes of providing, directly or indirectly, financing in connection with a transaction in which the Trust is the vendor or with the intention of using such mortgage as part of a method for subsequently acquiring an interest in or control of a property or a portfolio of properties that would otherwise meet the investment guidelines of the Trust; provided that the aggregate value of the investments of the

Trust in these mortgages, after giving effect to the proposed investments, will not exceed 20% of the Gross Book Value; and

(j) the Trust may invest, from time to time, an amount (which, in the case of an amount invested to acquire real property, is the purchase price less the amount of any indebtedness assumed or incurred by the Trust and secured by a mortgage on such property) up to 15% of the Adjusted Unitholders' Equity in investments which do not comply with one or more of subsections 4.1(d) or (g).

For the purpose of the foregoing guidelines, the assets, liabilities and transactions of a corporation or other entity wholly or partially owned by the Trust will be deemed to be those of the Trust on a proportionate consolidated basis. In addition, any references in the foregoing to investment in real property will be deemed to include an investment in a joint venture entity. For greater certainty, Sections 4.1(a) through (j) are intended to set out generally the parameters under which subsidiaries in which the Trust is permitted to invest will be empowered under their constating documents to reinvest. Further, any determinations in respect of the investment restrictions that are determinations reserved to the Trustees, where the actual activity is carried on by a subsidiary, will be made by the trustees or directors of the relevant subsidiary. Nothing in Section 4.1(a) and paragraphs (c) through (j) empowers or entitles the Trust or the Trustees to carry on business or to otherwise undertake any activity that would violate Section 4.3.

4.2 Operating Policies

The operations and affairs of the Trust shall be conducted in accordance with the following policies:

- (a) the Trust shall not purchase, sell, market or trade in currency or interest rate futures contracts otherwise than for hedging purposes where, for the purposes hereof; the term "hedging" shall have the meaning ascribed thereto by National Instrument 81-102 adopted by the Canadian Securities Administrators, as amended from time to time;
- (b) (i) any written instrument creating an obligation which is or includes the granting by the Trust of a mortgage, and (ii) to the extent management of the Trust determines to be practicable, any written instrument which is, in the judgment of management of the Trust, a material obligation, shall contain a provision or be subject to an acknowledgement in the form substantially provided for in Section 10.10;
- (c) the Trust may engage (i) in construction or development of real property in order to maintain its real properties in good repair and/or to enhance the income-producing potential of properties in which the Trust has an interest; and (ii) in the development of raw land and/or other development projects, provided that investments by the Trust in such development are within the investment guidelines set out in Section 4.1;
- (d) title to each real property shall be held by and registered in the name of the Trust, the Trustees or in the name of a corporation or other entity owned, directly or indirectly, by the Trust or jointly-owned, directly or indirectly, by the Trust, with joint venturers

or a corporation which is a nominee of the Trust which holds registered title to such real property pursuant to a nominee agreement with the Trust;

- (e) the Trust may directly or indirectly guarantee indebtedness or liabilities of a third party, provided that such guarantee is related to the direct or indirect ownership or acquisition by the Trust of real property that would otherwise comply with the investment guidelines and operating policies contained in Sections 4.1 and 4.2;
- (f) the Trust will obtain an independent appraisal of each property that it intends to acquire;
- (g) the Trust shall obtain and maintain at all times insurance coverage in respect of potential liabilities of the Trust and the accidental loss of value of the assets of the Trust from risks, in amounts, with such insurers, and on such terms as the Trustees consider appropriate, taking into account all relevant factors including the practices of owners of comparable properties and the cost of such coverage; and
- (h) the Trust shall obtain or review a Phase I environmental audit of each real property to be acquired by it, dated within 12 months of the proposed date of acquisition, and, if the Phase I environmental audit report recommends or recommended a Phase II environmental audit be obtained, the Trust shall obtain or review a Phase II environmental audit, in each case prepared by an independent environmental consultant.

For the purpose of the foregoing policies, the assets, liabilities and transactions of a corporation or other entity wholly or partially owned by the Trust will be deemed to be those of the Trust on a proportionate consolidated basis. In addition, any references in the foregoing investment restrictions and operating policies to investment in real property will be deemed to include an investment in a joint venture entity. For greater certainty Sections 4.2(a) through (h) are intended to set out generally the parameters under which the Trust (and subsidiaries in which the Trust is permitted to invest) will operate. References to the Trust in those paragraphs shall be read as applying to such subsidiary where the actual activity that is the subject of the policy is carried on by such subsidiary (with the exception of Section 4.2(b)(ii) and Section 4.2(f), which are only intended to apply to the Trust. Further, any determinations in respect of the operating policies that are determinations reserved to the Trustees, where the actual activity is carried on by a subsidiary, will be made by the trustees or directors of the relevant subsidiary. Nothing in Sections 4.2(a) through (h) empowers or entitles the Trust or the Trustees to carry on business or otherwise undertake any activity that would violate Section 4.3.

4.3 Tax Status

The Trustees shall cause the Trust to elect, in its return of income for the first taxation year of the Trust, pursuant to Subsection 132(6.1) of the Tax Act, that the Trust be deemed to be a "mutual fund trust" for the purposes of the Tax Act from the date it was established, provided that the Trust has sufficient unitholders so as to be entitled to make such election. Notwithstanding anything else contained in this Declaration of Trust, the Trust shall not make any investment, take any action or omit to take any action that would result in the Trust failing or ceasing to qualify as a "mutual fund

trust" within the meaning of the Tax Act and the Trust shall use its reasonable best efforts not to be a SIFT Trust.

4.4 Application of Investment Guidelines and Operating Guidelines

With respect to the investment guidelines and operating guidelines contained in Sections 4.1 and 4.2, where any maximum or minimum percentage limitation is specified in any of the restrictions therein contained, such restrictions shall be applied on the basis of the relevant amounts calculated immediately after the making of such investment or incurring or assuming the indebtedness. Any subsequent change relative to any percentage limitation which results from a subsequent change in the book value or appraised value of the assets of the Trust, the Adjusted Unitholders' Equity, or the Gross Book Value will not require divestiture of any investment or reduction of any indebtedness.

4.5 Regulatory Matters

If at any time a regulatory authority having jurisdiction over the Trust or any property of the Trust shall enact any law, regulation or requirement which is in conflict with any investment guideline of the Trust then in force, such guideline in conflict shall, if the Trustees so resolve, be deemed to have been amended to the extent necessary to resolve any such conflict, and, notwithstanding anything to the contrary herein contained, any such resolution of the Trustees shall not require the prior approval of Voting Unitholders.

ARTICLE 5 DISTRIBUTIONS

5.1 Computation of Income and Net Realized Capital Gains

The income of the Trust (the "**Trust Income**") for any taxation year of the Trust will be the income for such year computed in accordance with the provisions of the Tax Act, other than paragraph 82(1)(b) and subsection 104(6) thereof, regarding the calculation of income for the purposes of determining the "**taxable income**" of the Trust; provided, however, that capital gains and capital losses will be excluded from the computation of Trust Income and, if an amount has been designated by the Trust under subsection 104(19) of the Tax Act, such designation shall be disregarded.

The net realized capital gains of the Trust (the "Net Realized Capital Gains") for any taxation year of the Trust will be determined as the amount, if any, by which the aggregate of the capital gains of the Trust realized in such year exceeds the sum of (i) the aggregate of the capital losses of the Trust realized in such year, (ii) any capital gains that are realized by the Trust as a result of a redemption of Units pursuant to Article 6 and that are payable to redeeming Unitholders pursuant to Section 6.5, and (iii) each amount determined by the Trustees in respect of any net capital loss for a prior taxation year that the Trust is permitted by the Tax Act to deduct in computing the taxable income of the Trust for such year multiplied by the reciprocal of the fraction applicable in determining the proportion of a capital gain for such prior taxation year that is a taxable capital gain.

5.2 Distributions

- (1) The Trustees may, in respect of each Distribution Period, on or before each Distribution Record Date, declare payable to the Unitholders of record at the close of business on each Distribution Record Date, all or any part of the cash flow of the Trust for the Distribution Period. The amount of the cash flow of the Trust (being the sum of all cash amounts received by the Trust in respect of such Distribution Period, other than, for greater certainty, the proceeds of any offering of Units) to be distributed by the Trust in respect of a Distribution Period (the "**Distributions**") will be determined by, or in accordance with guidelines established from time to time by, the Trustees on or before the applicable Distribution Payment Date.
- (2) Distributions will be payable to each Unitholder of record on such Distribution Record Date pro rata in proportion to the number of Units held as of record by such Unitholder on such Distribution Record Date. Subject to Section 5.6, Distributions that have been declared to be payable to Unitholders in respect of a Distribution Period will be paid in cash on the Distribution Payment Date in respect of such Distribution Period.

5.3 Other Distributions

- (1) In addition to the Distributions that are declared payable to Unitholders pursuant to Section 5.2, the Trustees may allocate, declare payable and/or make distributions, from time to time, out of Trust Income, Net Realized Capital Gains, the capital of the Trust or otherwise, in any year, in such amount or amounts, and on such dates and to Unitholders of record on such dates, as the Trustees may determine.
- Having regard to the present intention of the Trustees to allocate, distribute and make payable to Unitholders all of the Trust Income, a sufficient amount of the Net Realized Capital Gains and any other applicable amounts so that the Trust will not have any liability for tax under Part I of the Tax Act in any taxation year (other than tax on Net Realized Capital Gains that would be recoverable by it with respect to the relevant taxation year or, in the event that the Trust is not a mutual fund trust at any time in the relevant taxation year, tax under Division E.1 of Part I of the Tax Act) (the "taxation distribution amount"), the following amounts will, unless the Trustees in their absolute discretion have otherwise determined to not distribute such taxation distribution amount but in lieu thereof to distribute another specified amount, be due and payable to Unitholders of record at the close of business on December 31 in each year:
 - (a) the amount of Trust Income for such year not previously paid or made payable to Unitholders in such year, provided that if the Trust is a SIFT Trust, the aggregate amount so payable for the year shall be reduced by such amount as is necessary to provide for the tax payable by the Trust by virtue of the SIFT Proposals, including, in particular, proposed paragraph 104(6)(b)(iv) of the Tax Act or any similar provision; and
 - (b) the amount of Net Realized Capital Gains for such year not previously paid or made payable to Unitholders in such year, except to the extent of Net Realized Capital Gains in respect of which the tax payable by the Trust would be refunded as a "capital gains refund" as defined in the Tax Act for such year, provided that if the Trust is a SIFT Trust, the aggregate amount so payable for the year shall be reduced

by such amount as is necessary to provide for the tax payable by the Trust by virtue of the SIFT Proposals, including, in particular, proposed paragraph 104(6)(b)(iv) of the Tax Act or any similar provision.

For greater certainty, if the Trustees have exercised their absolute discretion to not distribute the taxation distribution amount in respect of any year, the difference between amounts actually declared as Distributions and the taxation distribution amount in respect of such year shall not be payable to Unitholders. Moreover, the Trustees, in their absolute discretion, may at any time refute the intention referred to above to distribute taxation distribution amounts in respect of any year or future year.

- (3) Any distribution made pursuant to this Section 5.3 will be payable to each Unitholder of record on the applicable record date in respect of a distribution pursuant to Section 5.3(1), or on December 31 in the year of distribution in respect of a distribution pursuant to Section 5.3(2), pro rata in proportion to the number of Units held of record by such Unitholder on such applicable record date or December 31 in the year of such distribution, as the case may be. Subject to Section 5.6, amounts that have been declared to be payable to Unitholders pursuant to Section 5.3(1) will be paid in cash on the Distribution Payment Date determined by the Trustees in respect of such distribution and amounts that are payable pursuant to Section 5.3(2) will be paid in cash or Units not later than January 31 of the following year.
- (4) As contemplated in Section 6.5, the Trustees may designate as payable to redeeming Unitholders as part of the redemption price any capital gain and/or income realized by the Trust as a result of the redemption of Units pursuant to Section 6.5.

5.4 Character of Distributions and Designations

In accordance with and to the extent permitted by the Tax Act, the Trustees in each year shall make designations for income tax purposes in respect of the amounts paid or payable or deemed to be paid to Unitholders for such amounts that the Trustees consider to be reasonable in all of the circumstances, including designations relating to taxable dividends received by the Trust in the year on shares of taxable Canadian corporations, net capital gains realized by the Trust in the year and foreign source income of the Trust for the year, as well as elect under subsection 104(13.1) and/or subsection 104(13.2) of the Tax Act that income be taxed to the Trust, rather than to the Unitholders. Distributions paid or payable to Unitholders pursuant to this Article 5 will be deemed to be distributions of Trust Income, Net Realized Capital Gains, trust capital or other items in such amounts as the Trustees may, in their absolute discretion, determine. For greater certainty, it is hereby declared that any distribution of Net Realized Capital Gains will include the non-taxable portion of the capital gains of the Trust that are encompassed in such distribution.

5.5 Enforceability of Right to Receive Distributions

Notwithstanding any other provision of this Article 5, each Unitholder will have the legal right to enforce payment on the Distribution Payment Date or December 31, as the case may be, of any amount payable to such Unitholder as a result of any distribution declared or otherwise made payable pursuant to this Article 5 on the applicable Distribution Record Date or the applicable December 31, as the case may be, to, and not yet received by, such Unitholder pursuant to this Article 5.

5.6 Method of Payment of Distributions

- (1) Where the Trustees determine that the Trust does not have available cash in an amount sufficient to make payment of the full amount of any distribution that has been declared payable, or otherwise made payable, pursuant to this Article on the due date for such payment, the payment may, at the option of the Trustees, include or consist entirely of the issuance of additional Units, or fractions of Units, if necessary or desirable, having a value equal to the difference between the amount of such distribution declared to be payable and the amount of cash that has been determined by the Trustees to be available for the payment of such distribution. Such additional Units will be issued pursuant to applicable exemptions under applicable securities laws, discretionary exemptions granted by applicable securities regulatory authorities or a prospectus or similar filing.
- The value of each Unit that is issued pursuant to Section 5.6(1) will be the Market Price of the Units on the principal market on which the Units are quoted for trading during the trading day period ending on the trading day immediately prior to (i) the applicable Distribution Record Date in respect of a distribution pursuant to Section 5.2, (ii) the applicable record date in respect of a distribution under Section 5.3(1), or (iii) December 31 in respect of a distribution under Section 5.3(2).

5.7 Withholding Taxes

The Trustees shall deduct or withhold from distributions payable to any Unitholder all amounts required by law to be withheld from such distributions, whether such distributions are in the form of cash or otherwise. In the event of a distribution in the form of additional Units, the Trustees may sell Units of such Unitholder to pay such withholding taxes and to pay all of the Trustees' reasonable expenses with regard thereto and the Trustees shall have the power of attorney of such Unitholder to do so. Any such sale shall be made on any stock exchange on which the Units are then listed and upon such sale, the affected Unitholder shall cease to be the holder of such Units.

5.8 Definitions

Unless otherwise specified or the context otherwise requires, any term in this Article that is defined in the Tax Act will have for the purposes of this Article the meaning that it has in the Tax Act.

5.9 Payments of Cash

Any payment of cash by the Trust to a Unitholder pursuant to this Article 5 or any other provision of this Declaration of Trust will be conclusively deemed to have been made upon receipt of a wire transfer or other means of electronic funds transfer or upon mailing of a cheque in a postage pre-paid envelope, addressed to the Unitholder at the Unitholder's address appearing in the Register, unless such cheque is dishonored upon presentment. Upon such payment, the Trust will be discharged from all liability to the Unitholder in respect of such payment; provided, however, that if such cheque is lost or destroyed then, upon the presentation of evidence satisfactory to the Trustees of such loss or destruction, together with such indemnity as the Trustees may reasonably require, the Trust will issue a replacement cheque to the Unitholder.

5.10 Distribution Reinvestment and Unit Purchase Plan

Subject to any required regulatory approvals (and any Voting Unitholder approval imposed by regulatory requirements), the Trustees may, in their sole discretion, establish one or more Unitholder rights plans, distribution reinvestment plans and Unit purchase plans or Unit option plans, including the Unit Option Plan, at any time and from time to time.

5.11 Income Tax Matters

In computing the net income of the Trust for income tax purposes for any year, except as the Trustees otherwise determine, the Trust shall claim the maximum amount of capital cost allowance and other discretionary deductions available to the Trust under the Tax Act.

ARTICLE 6 REDEMPTION OF UNITS

6.1 Redemption Right

Each Unitholder will be entitled to require the Trust to redeem at any time or from time to time at the demand of the Unitholder all or any part of the Units registered in the name of the Unitholder at the prices determined and payable in accordance with this Article 6.

6.2 Exercise of Redemption Right

- (1) Units may be surrendered at any time to the Transfer Agent for redemption by presentation to the Transfer Agent and the Trust of a written notice of redemption from the Unitholder or CDS, as the case may be, in a form acceptable to the Transfer Agent and approved by the Trustees, specifying the number of Units to be redeemed and accompanied by the Unit Certificate(s) representing the Units to be redeemed if a Unit Certificate has been issued. Any expense associated with the preparation and delivery of redemption notices will be for the account of the Unitholder exercising the redemption privilege. Such surrender will be irrevocable except with respect to any Units surrendered for redemption in respect of which the redemption proceeds are not paid by the Trust on or before the date on which such payment is due and except as otherwise provided herein.
- desires to redeem Units shall do so by causing a CDS Participant to deliver to CDS on behalf of the Beneficial Unitholder a written notice of the Beneficial Unitholder's intention to redeem Units, which notice shall then be provided to the Transfer Agent and the Trust. The form of redemption notice shall be substantially in the form set forth on the Unit Certificate. By causing a CDS Participant to deliver to CDS a notice of the Beneficial Unitholder's intention to redeem Units, a Beneficial Unitholder will be deemed to have irrevocably surrendered the applicable Units for redemption and appointed such CDS Participant to act as his or her exclusive settlement agent with respect to the exercise of the redemption privilege and the receipt of payment in connection with the settlement of obligations arising from such exercise. Any redemption notice that CDS determines to be incomplete, not in proper form or not duly executed will for all purposes be void and of no effect, and the redemption privilege to which it relates will be considered for all purposes not to have been exercised thereby. A failure by CDS or a CDS Participant to exercise redemption privileges or to

give effect to the settlement thereof in accordance with the Beneficial Unitholder's instructions will not give rise to any obligations or liability on the part of the Trust or the Trustees to the CDS Participant or to the Beneficial Unitholder. A Beneficial Unitholder will have no direct redemption rights as against the Trust or the Trustees, and any such redemption effected by such Beneficial Unitholder will, insofar as the Trust or the Trustees are concerned, be deemed to be on the part of the registered Unitholder.

(3) As of the close of business on the date when the Units are surrendered for redemption, the Unitholder will cease to be the Unitholder of record in respect of such Units, such Units will cease to be entitled to share in the income or any participation in the assets of the Trust (other than the receipt of the redemption proceeds), and the Unitholder (or Beneficial Unitholder) thereof will not be entitled to exercise any of the rights of holders of Units in respect thereof.

6.3 Cash Redemption

- (1) Subject to Section 6.4, upon receipt by the Trust of a notice to redeem Units in accordance with Section 6.2 and surrender of the relevant Unit Certificate (as applicable), the Unitholder or Beneficial Unitholder of the Units tendered for redemption will, subject to Section 6.5, be entitled to receive a price per Unit (the "**Redemption Price**") equal to the lesser of:
 - (a) 90% of the Market Price of the Units on the principal stock exchange on which the Units are listed (or, if the Units are not listed on any stock exchange, on the principal market on which the Units are quoted for trading) during the 10 trading day period ending immediately prior to the date on which the Units were surrendered to the Trust for redemption; and
 - (b) the Closing Market Price of the Units on the date on which the Units were surrendered to the Trust for redemption on the principal stock exchange on which the Units are listed (or, if the Units are not listed on any stock exchange, on the principal market on which the Units are quoted for trading).

For the purposes of this Declaration of Trust, "Market Price" will be the amount equal to the weighted average of the trading prices of the Units on the applicable market or exchange for each of the trading days on which there was a trade during the specified trading day period; provided that if there was trading on the applicable exchange or market for fewer than five of the trading days during the specified trading day period, "Market Price" will be the average of the following prices established for each of the trading days during the specified trading day period: the average of the last bid and last ask prices for each trading day on which there was no trading and the weighted average trading prices of the Units for each trading day that there was trading. For the purposes of this Declaration of Trust, "Closing Market Price" will be: (i) an amount equal to the closing price of the Units on the applicable market or exchange if there was a trade on the specified date and the applicable market or exchange provides a closing price; (ii) an amount equal to the average of the highest and lowest prices of Units on the applicable market or exchange if there was trading on the specified date and the applicable market or exchange provides only the highest and lowest trading prices of Units traded on a particular day; or (iii) the average of the last bid and last ask prices on the applicable market or exchange if there was no trading on the specified date.

Units surrendered for redemption during any calendar month will be satisfied by way of cash payment no later than the last day of the calendar month following the calendar month in which the Units were tendered for redemption. Payments made by the Trust of the cash portion of the Redemption Price will be conclusively deemed to have been made upon the mailing by the Trust of a cheque in a postage-prepaid envelope addressed to the former Unitholder at the former Unitholder's address appearing in the Register, or as otherwise advised by a CDS Participant in the event of a redemption by a Beneficial Unitholder, unless such cheque is dishonored upon presentment. Upon such payment, the Trust will be discharged from all liability to the former Unitholder (or Beneficial Unitholder) in respect of the redeemed Units; provided, however, that if such cheque is lost or destroyed then, upon the presentation of evidence satisfactory to the Trustees of such loss or destruction, together with such indemnity as the Trustees may reasonably require, the Trust will issue a replacement cheque to the former Unitholder (or Beneficial Unitholder).

6.4 No Cash Redemption in Certain Circumstances

Section 6.3(2) will not be applicable to Units tendered for redemption by a Unitholder, if:

- (a) the total amount payable by the Trust pursuant to Section 6.3 in respect of such Units and all other Units tendered for redemption in the same calendar month exceeds \$50,000 (the "Monthly Limit"), provided that the Trustees may, in their sole discretion, waive such limitation in respect of all Units tendered for redemption in any calendar month. Units tendered for redemption in any calendar month in which the total amount payable by the Trust pursuant to Section 6.3(2) exceeds the Monthly Limit will be redeemed for cash pursuant to Section 6.3(2) and, unless any applicable regulatory approvals are required, by a distribution in specie under Section 6.5, on a pro rata basis;
- (b) on the date the Units are tendered for redemption, the outstanding Units are not listed for trading or quoted on any stock exchange or market that the Trustees consider, in their sole discretion, to provide a representative fair market value price for the Units;
- on the date the Units are tendered for redemption or if, for more than five trading days during the 10 trading day period immediately prior to the date on which such Units were tendered for redemption, the normal trading of the outstanding Units is suspended or halted on any stock exchange on which the Units are listed for trading or, if not so listed, on any market on which the Units are quoted for trading; or
- (d) the redemption of the Units will result in the delisting of the Units on the principal stock exchange on which the Units are listed.

6.5 In Specie Redemption

If, pursuant to Section 6.4(a), Section 6.3(2) is not applicable to all or any of the Units tendered for redemption by a Unitholder, then such holder of Units shall, instead of the Redemption Price per Unit specified in Section 6.3 payable in respect of such Units, to be entitled to receive a price per Unit (the "In specie Redemption Price") equal to the fair market value of a Unit as determined by the Trustees, and the In specie Redemption Price shall, subject to all necessary

regulatory approvals, be paid and satisfied by the Trust issuing Redemption Notes having an aggregate principal amount equal to the aggregate In specie Redemption Price of the Units tendered for redemption. In the event of distributions of Redemption Notes, each Redemption Note so distributed to the redeeming holder of Units shall be in the principal amount of \$100 or such other amount as may be determined by the Trustees. No fractional Redemption Notes shall be distributed and where the number of Redemption Notes to be received upon redemption by a holder of Units would otherwise include a fraction, that number shall be rounded down to the next lowest whole number.

6.6 Cancellation of all Redeemed Units

All Units that are redeemed under this Article 6 will be cancelled, and such Units will no longer be outstanding and will not be reissued.

6.7 Withholdings by the Trustees

The Trustees may deduct or withhold from all payments or other distributions payable to any Unitholder pursuant to this Article 6 all amounts required by law to be so withheld.

6.8 Redemption of Initial Unit

If at any time the Unit held by the Initial Unitholder is the only outstanding Unit, the holder thereof may require the Trust to redeem the Unit for an amount equal to the Trust Assets.

ARTICLE 7 REDEMPTION OF SPECIAL VOTING UNITS

7.1 Right of Redemption

Each holder of Special Voting Units will be entitled to require the Trust to redeem, at any time and from time to time, at the demand of such holder, all or any part of the Special Voting Units registered in the name of such holder at the prices determined and payable in accordance with the provisions of this Article 7.

7.2 Exercise of Redemption Right

Special Voting Units may be surrendered at any time to the Transfer Agent for redemption by presentation to the Trust of a written request for redemption from the registered holder of the Special Voting Units in a form acceptable to the Trust, specifying the number of Special Voting Units to be redeemed. Such surrender will be irrevocable except with respect to any Special Voting Units surrendered for redemption in respect of which the redemption proceeds are not paid by the Trust on or before the date on which such payment is due and except as otherwise provided herein.

As of the close of business on the date the Special Voting Units are surrendered for redemption, the registered holder of the Special Voting Units will cease to be the holder of record in respect of such Special Voting Units, and will cease to have any rights with respect to the Special Voting Units tendered for redemption (other than the receipt of the redemption proceeds).

7.3 Cash Redemption

Upon receipt by the Trust of a notice to redeem Special Voting Units in accordance with Section 7.2, the holder of the Special Voting Units tendered for redemption will be entitled to receive a price per Special Voting Unit equal to \$0.00001. All payments made in respect of the redemption of Special Voting Units will be rounded up to the nearest whole cent.

The redemption price payable in respect of the Special Voting Units surrendered for redemption during any calendar month will be satisfied by way of cash payment no later than the last day of the calendar month following the calendar month in which the Special Voting Units were tendered for redemption. Payments made by the Trust of the cash portion of the redemption price for such Special Voting Units will be conclusively deemed to have been made upon the mailing by the Trust of a cheque in a postage-prepaid envelope addressed to the former holder of the Special Voting Units at the former holder's address appearing in the Register, unless such cheque is dishonored upon presentment. Upon such payment, the Trust will be discharged from all liability to the former holder of the Special Voting Units in respect of the redeemed Special Voting Units; provided, however, that if such cheque is lost or destroyed then, upon the presentation of evidence satisfactory to the Trustees of such loss or destruction, together with such indemnity as the Trustees may reasonably require, the Trust will issue a replacement cheque to the former holder of the Special Voting Units.

7.4 Cancellation of all Redeemed Special Voting Units

All Special Voting Units that are redeemed under this Article 7 will be cancelled, and such Special Voting Units will no longer be outstanding and will not be reissued.

ARTICLE 8 TRUSTEES

8.1 Number of Trustees

The Trust shall have not less than three (3) and no more than 11 Trustees, with the number of Trustees from time to time within such range being fixed (the "**Fixed Number**") by resolution of the Trustees; provided that until otherwise so determined by resolution, the number of Trustees will be five (5). A majority of the Trustees shall be Resident Canadians and shall be Independent Trustees.

8.2 Calling and Notice of Meetings

Meetings of the Trustees will be called and held from time to time at such time and at such place as the Trustees, the Chair of the Trustees or any two Trustees may determine, and any one Trustee or officer of the Trust may give notice of meetings when directed or authorized by such persons. Notice of each meeting of the Trustees will be given to each Trustee not less than 48 hours before the time when the meeting is to be held, provided that if a quorum of Trustees is present, the Trustees may without notice hold a meeting immediately following an annual meeting of Voting Unitholders. A Trustee may waive this notice and the presence of such Trustee at such a meeting will be deemed to be a waiver of this notice requirement except where such Trustee attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting has not been lawfully called or convened. Notice of a meeting of the Trustees may be given

verbally, in writing or by telephone, facsimile or electronic mail. A notice of a meeting of Trustees need not specify the purpose of, or the business to be transacted at, the meeting. Notwithstanding the foregoing, the Trustees may by resolution from time to time fix a day or days in any month or months for regular meetings of the Trustees at a place and hour to be named, in which case, provided that a copy of such resolution is sent to each Trustee forthwith after being passed and forthwith after each Trustee's appointment, no other notice will be required for any such regular meeting.

8.3 Place of Meetings

Meetings of the Trustees may be held at any place in Canada. A Trustee who attends a meeting of Trustees, in person or by telephone, is deemed to have consented to the location of the meeting except when he or she attends the meeting for the express purpose of objecting to the transaction of any business on the grounds that the meeting is not lawfully held.

8.4 Meetings by Telephone

With the consent of the chair of the meeting or a majority of the Trustees participating in the meeting, any Trustee may participate in a meeting of the Trustees or of a committee of the Trustees from any location in the world by means of telephone, electronic or other communication facilities that permit all persons participating in the meeting to communicate with each other. A Trustee participating in such a meeting in such manner will be considered present at the meeting and at the place of the meeting. Any such consent will be effective whether given before or after the meeting to which it relates and may be given with respect to all meetings of the Trustees and of committees of Trustees.

8.5 Quorum

The quorum for the transaction of business at any meeting of the Trustees will consist of the greater of two Trustees and a majority of the number of Trustees then holding office (provided that the majority of Trustees present are Resident Canadians), and, notwithstanding any vacancy among the number of Trustees, a quorum of Trustees may exercise all of the powers of the Trustees.

8.6 Chair

The chair of any meeting of the Trustees will be the Trustee present at the meeting who holds the office of Chair of the Trustees (as appointed by the Trustees under Section 8.10) or, if such person is not present at the meeting, the Trustees present at the meeting shall choose one of their number to be chair.

8.7 Action by the Trustees

Every question at all meetings of the Trustees will be decided by a majority of the votes cast on the question. In the case of an equality of votes, the chair of the meeting will not be entitled to a second or casting vote. The powers of the Trustees may be exercised by resolution passed at a meeting at which a quorum is present or by a resolution in writing signed by all Trustees who would be entitled to vote on that resolution at a meeting of the Trustees. Resolutions in writing may be signed in counterparts (including by facsimile or in electronic format), each of which will be deemed to be an original and all originals together will be deemed to be one and the same instrument.

8.8 Adjourned Meeting

Any meeting of Trustees may be adjourned from time to time by the chair of the meeting with the consent of the meeting to a fixed time and place. Further notice of the adjourned meeting need not be given. The adjourned meeting will be duly constituted if a quorum is present and if it is held in accordance with the terms of the adjournment. If there is not a quorum present at the adjourned meeting, the original meeting will be deemed to have terminated upon its adjournment.

8.9 Remuneration and Expenses

The Trustees shall be paid such remuneration for their services as the Trustees, or if applicable a committee thereof, may from time to time determine. The Trustees shall also be entitled to be reimbursed for reasonable traveling and other expenses properly incurred by them in attending meetings of the Trustees or any committee thereof or in connection with their services as Trustees. Nothing herein contained shall preclude any Trustee from serving the Trust in any other capacity and receiving remuneration therefor.

8.10 Officers

The Trustees may from time to time appoint one or more officers of the Trust, including without limitation, a Chair and a Secretary of the Trustees, provided that:

- (1) no person shall be deemed to be non-independent under applicable securities laws by reason only of his or her appointment as Chair and/or Secretary of the Trustees); and
- (2) unless the Independent Trustees otherwise unanimously determine, the Chair shall be an Independent Trustee,

and, without prejudice to rights under any employment contract with the Trust and/or the Manager, may remove any officer of the Trust. The powers and duties of each officer of the Trust will be determined from time to time by the Trustees and, in the absence of such determination, will be those usually applicable to the office held.

8.11 Committees

Except as prohibited by law, the Trustees may appoint from their number a committee of Trustees and may delegate to the committee of Trustees such authority as the Trustees may in their sole discretion deem necessary or desirable to effect the administration of the duties of the Trustees under this Declaration of Trust, without regard to whether such authority is normally granted or delegated by trustees. A majority of the Trustees appointed to any committee shall be Resident Canadians and, with the exception of the Audit Committee, Independent Trustees. All Trustees appointed to the Audit Committee shall be Independent Trustees.

8.12 Committee Procedure

Unless otherwise determined by the Trustees, a quorum for meetings of any committee will be a majority of its members (provided that a majority of the Trustees comprising such quorum shall be Resident Canadians) and each committee will have the power to appoint its chair, and the rules for calling, holding, conducting, transacting business at and adjourning meetings of the committee

will be the same as those governing the Trustees. Each member of a committee will serve during the pleasure of the Trustees and, in any event, only so long as he or she is a Trustee. The Trustees may fill vacancies in a committee by appointment from among their members. Provided that a quorum is maintained, the committee may continue to exercise its powers notwithstanding any vacancy among its members.

ARTICLE 9 APPOINTMENT, RESIGNATION AND REMOVAL OF THE TRUSTEES

9.1 Qualification of Trustees

The following persons are disqualified from being a Trustee of the Trust:

- (a) anyone who is less than 18 years of age;
- (b) anyone who is of unsound mind and has been so found by a court of competent jurisdiction in Canada or elsewhere;
- (c) a person who is not an individual; or
- (d) a person who has the status of bankrupt.

9.2 Proposed Manager Trustees

- So long as the Management Agreement is in effect, prior to each annual meeting of Voting Unitholders and any special meeting of Voting Unitholders at which Trustees are to be elected and so long as the number of Trustees to be elected is five, the Manager shall be entitled to designate two individuals (the "Proposed Manager Trustees") (which may include the reappointment of incumbent Trustees) to stand for election as Trustees at such meeting. The designation of the Proposed Manager Trustees shall be made by written notice (the "Proposed Manager Trustee Notice") signed by an authorized signatory of the Manager to the Chair of the Trustees. The Manager shall ensure prior to the designation of a Proposed Manager Trustee that such individual is qualified to act as a Trustee pursuant to the requirements of the Declaration of Trust, and that the Proposed Manager Trustees, individually and collectively, meet all applicable legal and other requirements respecting the qualifications of individuals acting as Trustees, including, without limitation, with respect to residency, financial literacy and independence. Upon notification by the Trust that any Proposed Manager Trustee is not qualified to act as a Trustee, the Manager shall forthwith designate a substitute Trustee who is qualified to act as a Trustee. Notwithstanding anything else contained herein, if the Manager fails to provide the Proposed Manager Trustee Notice to the Chair within five business days after the Record Date of the meeting at which the Trustees will stand for election, the existing Trustees shall designate the Proposed Manager Trustees (which may include the reappointment of incumbent Trustees).
- (2) The number of Proposed Manager Trustees designated by the Manager pursuant to Section 9.2(1) shall be adjusted proportionately to reflect any increase or decrease in the number of Trustees. In the event that any such adjustment results in a fractional number of Proposed Manager Trustees to be designated by the Manager, such fraction shall be rounded upwards to the closest whole number unless this would cause the number of Proposed Manager Trustees to constitute more

than 49% of all of the Trustees in which case such fractional number shall be rounded downward to the closest whole number.

9.3 Election of Trustees

- (1) The appointment of the initial Trustees is hereby confirmed, such appointment to be for an initial term of office that, subject to Section 9.6, will expire (subject to further appointment) at the close of the first annual meeting of Voting Unitholders. The Trustees may, between annual meetings of Voting Unitholders, and subject to Section 8.1, appoint one or more additional Trustees to serve until the next annual meeting of Voting Unitholders, but the number of additional Trustees so appointed may not at any time exceed the greater of:
 - (a) two Trustees at any time that there are five or more Trustees; and
 - (b) one-third of the number of Trustees who held office at the expiration of the immediately preceding annual meeting of Voting Unitholders,

provided that (i) any Trustees appointed pursuant to this Section are Independent Trustees and (ii) the determination to increase the number of Trustees and the decision as to who to appoint as a Trustee is made solely by the then Independent Trustees, who are also the only Trustees eligible to vote on the matter.

(2) Except as otherwise provided herein, Trustees will be elected (including the reelection of incumbent Trustees) at each annual meeting of Voting Unitholders, and may be elected at a special meeting of Voting Unitholders, in each case to hold office, subject to Section 9.6, for a term expiring at the close of the next annual meeting of Voting Unitholders following such election. Any such election (other than by the Trustees as contemplated by Sections 9.3 and 9.8) will be made by Ordinary Resolution. Notwithstanding anything else contained herein, if no Trustees are elected at the annual meeting of Voting Unitholders held immediately before the term of office of the existing Trustees expires, such existing Trustees will continue to be Trustees under this Declaration of Trust until their successors have been appointed or they otherwise cease to be Trustees.

9.4 Consent to Act

(1) A person who is elected or appointed as a Trustee hereunder, other than the initial Trustees whose consent to act is given by their signatures hereto, will not become a Trustee until such person has, either before or after such election or appointment, consented in writing to do so. Without limiting the form of such consent, the execution and delivery to the Trust of a consent substantially as follows will satisfy such requirement:

"To: Partners Real Estate Investment Trust (the "Trust")

And to: The Trustees thereof

The undersigned hereby consents to act as a Trustee of the Trust and hereby agrees, upon the earlier of the date of this consent and the date of the undersigned's appointment and/or election as a Trustee of the Trust, to become a party, as a Trustee, to the Declaration of Trust, as amended from time to time, constituting the Trust and to be bound by the obligations and liabilities of a Trustee thereunder.

Dated:		
	[Signature]	
	[Print Name]"	

- (2) Except with respect to the initial Trustees, upon the later of a person being elected or appointed a Trustee hereunder and executing and delivering to the Trust a consent substantially as set forth in Section 9.4(1), such person shall become a Trustee hereunder and shall be deemed to be a party (as a Trustee) to this Declaration of Trust, as amended and restated from time to time.
- Trust and have title in and to the property of the Trust drawn up in their names and all other rights to the Trustees at law shall vest automatically in all individuals who may hereafter become Trustees upon their becoming a Trustee in accordance with the provisions of this Declaration of Trust without any further act, and they shall thereupon have all the rights, privileges, powers, authorities, obligations and immunities of Trustees hereunder, whether or not conveyancing documents have been executed and delivered pursuant to Section 9.6 or otherwise.

9.5 Failure to Elect Minimum Number of Trustees

If a meeting of Voting Unitholders fails to elect the minimum number of Trustees required by this Declaration of Trust by reason of the disqualification or death of any nominee, the Trustees elected at the meeting may exercise all of the powers of the Trustees if the number of Trustees so elected constitutes a quorum.

9.6 Ceasing to Hold Office

- (1) A Trustee ceases to hold office when:
- (a) he or she dies or resigns;
- (b) he or she is removed in accordance with Section 9.7; or
- (c) he or she ceases to be duly qualified to act as a Trustee as provided under Section 9.1.
- (2) A resignation of a Trustee becomes effective at the time when a written resignation is received by the Trust, or at the time specified in the resignation, whichever is later.
- (3) Upon the resignation or removal of any Trustee, or upon a Trustee otherwise ceasing to be a Trustee, the Trustee will cease to have the rights, privileges and powers of a Trustee hereunder and will cease to be a party (as a Trustee) to this Declaration of Trust; provided, however, that such Trustee will continue to be entitled to be paid any amounts owing by the Trust to the Trustee and to the benefits of the indemnity provided in Section 10.9. Upon the resignation or removal of any Trustee, or upon a Trustee otherwise ceasing to be a Trustee, the Trustee shall execute and deliver such documents as the remaining Trustees may require for the conveyance of

any Trust Assets held in that Trustee's name, shall account to the remaining Trustees as they may require for all property which that Trustee holds as Trustee, shall resign from all representative or other positions held by such Trustee on behalf of the Trust, including as a director or officer of any person in which the Trust owns any securities (directly or indirectly), and will thereupon be discharged as Trustee.

(4) Upon the incapacity or death of any Trustee, his or her legal representative shall execute and deliver on his or her behalf such documents as the remaining Trustees may require under Section 9.6(3). In the event that a Trustee or his or her legal representatives, as applicable, are unable or unwilling to execute and deliver such required documents, each of the remaining Trustees is hereby appointed as the attorney of such Trustee for the purposes of executing and delivering such required documents. This power of attorney granted to each of the remaining Trustees is not intended to be a continuing power of attorney within the meaning of the *Substitute Decisions Act*, 1992 (Ontario), exercisable during a Trustee's incapacity to manage property, or any similar power of attorney under equivalent legislation in any of the provinces or territories of Canada (a "CPOA"). The execution of this power of attorney will not terminate any CPOA granted by the Trustee previously and will not be terminated by the execution by the Trustee in the future of a CPOA, and the Trustee hereby agrees not to take any action in future which results in the termination of this power of attorney.

9.7 Removal of Trustees

The Voting Unitholders may remove any Trustee or Trustees from office by an Ordinary Resolution passed at a meeting of Voting Unitholders called for that purpose. A vacancy created by the removal of a Trustee may be filled at the meeting of Voting Unitholders at which the Trustee is removed or, if not so filled, may be filled as set forth in Section 9.8.

9.8 Filling Vacancies

No vacancy among the Trustees shall operate to annul this Declaration of Trust or affect the continuity of the Trust. Until vacancies are filled, the remaining Trustee or Trustees (even if less than a quorum) may exercise the powers of the Trustees hereunder. A quorum of Trustees may fill a vacancy among the Trustees, except (i) a vacancy resulting from an increase in the number of Trustees other than as provided in Section 9.3(1), or (ii) from a failure to elect the minimum number of Trustees fixed by this Declaration of Trust. If there is not a quorum of Trustees, or if there has been a failure to elect the minimum number of Trustees required by or pursuant to this Declaration of Trust, the Trustees then in office will promptly call a special meeting of Voting Unitholders to fill the vacancy and, if they fail to so call a meeting or if there are no Trustees then in office, such meeting may be called by any Voting Unitholder. A Trustee appointed to fill a vacancy holds office, subject to Sections 9.6 and 9.7, until the close of the next annual meeting of the Voting Unitholders.

9.9 Validity of Acts

Any act of a Trustee is valid notwithstanding any irregularity in the election or appointment of all or any of the Trustees or a defect in the qualifications of all or any of the Trustees.

ARTICLE 10 CONCERNING THE TRUSTEES

10.1 Powers of the Trustees

Subject to the specific limitations contained in this Declaration of Trust, the Trustees shall have, without further or other action or consent, and free from any power or control on the part of the Voting Unitholders, full, absolute and exclusive power, control and authority over the Trust Assets and over the affairs of the Trust to the same extent as if the Trustees were the sole and absolute legal and beneficial owners of the Trust Assets in their own right, to do all such acts and things as in their sole judgment and discretion are necessary or incidental to, or desirable for, carrying out the trust created hereunder. In construing the provisions of this Declaration of Trust, the presumption shall be in favour of the granted powers and authority to the Trustees. The enumeration of any specific power or authority herein (including pursuant to Section 10.2) shall not be construed as limiting the general power or authority or any other specified power or authority conferred herein on the Trustees. To the maximum extent permitted by law the Trustees shall, in carrying out investment activities, not be in any way restricted by the provisions of the laws of any jurisdiction limited or purporting to limit investments which may be made by trustees. Without limiting the generality of the foregoing, but subject to Sections 4.1, 4.2 and 10.2 and any other express limitations contained in this Declaration of Trust, the Trustees may make any investments without being required to adhere to all of or any particular portion of the investment criteria or diversification requirements set forth in the *Trustee* Act (Ontario), as amended from time to time, including investments in mutual funds, common trust funds, unit trusts and similar types of investment vehicles, to alter or vary such investments from time to time in a like manner, to retain such investments for such length of time as the Trustees, in their discretion, determine and to delegate management and authority to discretionary managers of investment funds as the Trustees in their discretion determine appropriate.

10.2 Specific Powers and Authorities

Subject only to the terms and conditions contained in this Declaration of Trust and in addition to any other powers and authorities conferred by this Declaration of Trust or which the Trustees may have by virtue of any present or future statute or rule of law, the Trustees, without any action or consent by the Voting Unitholders, will have and may exercise at any time and from time to time any and all rights, powers and privileges that can be exercised by a legal and beneficial owner of the Trust Assets, including the following powers and authorities which may be exercised by the Trustees in such manner and upon such terms and conditions as they may from time to time deem proper:

- (a) to supervise the activities and manage the investments and affairs of the Trust;
- (b) to maintain records and provide reports to Voting Unitholders;
- (c) to determine the amount of Distributions and the Distribution Period;
- (d) to obtain security, including encumbrances on assets, to secure the full payment of monies owed to the Trust and the performance of all obligations in favour of the Trust, and to exercise all of the rights of the Trust, and to perform all of the obligations of the Trust, under such security;

- (e) to establish places of business of the Trust;
- (f) to do all such acts and things as are necessary to ensure that the Trust qualifies as a "unit trust" and a "mutual fund trust" pursuant to paragraph 108(2)(a) and subsection 132(6) of the Tax Act and, to the extent reasonably practicable, to undertake any reorganization of the Trust and its assets so as to ensure the Trust is not considered a SIFT Trust;
- (g) in addition to the mandatory indemnification provided for in Section 2.8 and Section 10.9, to indemnify, or enter into agreements with respect to the indemnification of, any person with whom the Trust has dealings, including without limitation the Trustees, any depositary, registrar, transfer agent or escrow agent (including CDS and the Transfer Agent), to such extent as the Trustees shall determine and to the extent permitted by law;
- (h) to grant broad discretion to a third party, including the Manager, to administer and manage the day-to-day operations of the Trust and to make executive decisions which conform to the general guidelines, policies and principles set forth in this Declaration of Trust or otherwise established by the Trustees from time to time;
- (i) to arrange for insurance contracts and policies insuring the Trust and its subsidiaries, Trust Assets (whether held by the Trust or its subsidiaries), and/or any or all of the Trustees, officers of the Trust or the Voting Unitholders, including against any and all claims and liabilities of any nature asserted by any person arising by reason of any action alleged to have been taken or omitted by the Trust or by the Trustees, officers of the Trust or Voting Unitholders;
- (j) to approve the adoption of a Unitholders rights plan, if the Trustees determine in good faith that such action is appropriate, including to issue rights in connection with such a plan;
- (k) to negotiate, make, execute, acknowledge and deliver any and all deeds, instruments, contracts, waivers, releases or other documents necessary or useful for the accomplishment of any of the powers herein granted, the purpose of the Trust or its assets or affairs, including, without limitation, the administration of the Trust, and to perform all of the obligations of the Trust thereunder;
- (l) to purchase Units or Voting Units for cancellation in accordance with applicable regulatory requirements;
- (m) to retain, invest and re-invest the capital or other funds of the Trust in real or personal property of any kind, all without regard to whether any such properties are authorized by law for the investment of trust funds, and to possess and exercise all the rights, powers and privileges appertaining to the ownership of the property of the Trust and to increase the capital of the Trust at any time by the issuance of additional Units for such consideration as they deem appropriate;

- (n) for such consideration as they deem proper, to invest in, purchase or otherwise acquire for cash or other property or through the issuance of Units or through the issuance of notes, debentures, bonds or other obligations or securities of the Trust and hold for investment the entire or any participating interest in any mortgages. In connection with any such investment, purchase or acquisition, the Trustees shall have the power to acquire a share of rents, lease payments or other gross income from or a share of the profits from or a share in the equity or ownership of real property;
- (o) to sell, rent, lease, hire, exchange, release, partition, assign, mortgage, pledge, hypothecate, grant security interests in, encumber, negotiate, convey, transfer or otherwise dispose of any or all of the property of the Trust by deeds, trust deeds, assignments, bills of sale, transfers, leases, mortgages, financing statements, security agreements and other instruments for any of such purposes executed and delivered for and on behalf of the Trust by one or more of the Trustees or by a duly authorized officer, employee, agent or any nominee of the Trust;
- (p) to enter into leases, contracts, obligations and other agreements for a term extending beyond the term of office of the Trustees and beyond the possible termination of the Trust or for a lesser term;
- (q) to borrow money from or incur indebtedness to any person; to guarantee, indemnify or act as surety with respect to payment or performance of obligations of third parties; to enter into other obligations on behalf of the Trust; and to assign, convey, transfer, mortgage, subordinate, pledge, grant security interests in, encumber or hypothecate the property of the Trust to secure any of the foregoing;
- (r) without limit as to amount, to issue any type of debt securities or convertible debt securities and to borrow money or incur any other form of indebtedness for the purpose of carrying out the purposes of the Trust or for other expenses incurred in connection with the Trust and for such purposes may draw, make, execute and issue promissory notes and other negotiable and non-negotiable instruments or securities and evidences of indebtedness, secure the payment of sums so borrowed or indebtedness incurred and mortgage, pledge, assign or grant a security interest in any money owing to the Trust or its property or engage in any other means of financing the Trust;
- (s) to lend money or other property of the Trust, whether secured or unsecured;
- (t) to incur and ratify or pay out of the property of the Trust any debts, claims, charges or expenses and disburse any funds of the Trust, which charges, expenses or disbursements are, in the opinion of the Trustees, necessary or incidental to or desirable for the carrying out of any of the purposes of the Trust or conducting the affairs of the Trust including, without limitation, taxes or other governmental levies, charges and assessments of whatever kind or nature, imposed upon or against the Trustees in connection with the Trust or the property of the Trust or upon or against the property of the Trust or any part thereof and for any of the purposes herein;

- (u) to deposit funds of the Trust in banks, trust companies and other depositories, whether or not such deposits will earn interest, the same to be subject to withdrawal on such terms and in such manner and by such person or persons (including, without limitation, any one or more Trustees, officers, agents or representatives) as the Trustees may determine;
- (v) to possess and exercise all the rights, powers and privileges appertaining to the ownership of or an interest in, all or any mortgages or securities, issued or created by any person, forming part of the assets of the Trust, to the same extent that an individual might and, without limiting the generality of the foregoing, to vote or give any consent, request or notice, or waive any notice, either in person or by proxy or power of attorney, with or without power of substitution, to one or more persons, which proxies and powers of attorney may be for meetings or action generally or for any particular meeting or action and may include the exercise of discretionary power;
- (w) to exercise any conversion privilege, subscription right, warrant or other right or option available in connection with any property of the Trust at any time held by it and to make payments incidental thereto; to consent, or otherwise participate in or dissent from, the reorganization, consolidation, amalgamation, merger or readjustment of the finances of any person (other than the Trust), any of the securities of which may at any time be held by the Trust or to the sale, mortgage or lease of the property of any such person; and to do any act with reference thereto, including (without limitation) the delegation of discretionary powers, the exercise of options, the making of agreements or subscriptions and the payment of expenses, assessments or subscriptions which it may consider necessary or advisable in connection therewith;
- to elect, appoint, engage or employ officers for the Trust (including, without (x) limitation, the Chair of the Trustees, Chief Executive Officer, President, Chief Financial Officer, non-executive Secretary and such vice-presidents and other officers as the Trustees may determine), who may be removed or discharged at the discretion of the Trustees, such officers to have such powers and duties, and to serve such terms as may be prescribed by the Trustees or by the Trustees' Regulations (as defined below); to engage, appoint, employ or contract with any persons as agents, representatives, employees or independent contractors or otherwise (including, without limitation, real estate advisors, investment advisors, registrars, underwriters, accountants, lawyers, real estate agents, property managers, appraisers, brokers, architects, engineers, construction managers, general contractors or otherwise) in one or more capacities, and to pay compensation from the Trust for services in as many capacities as such persons may be so engaged or employed; and, except as prohibited by law, to delegate any of the powers and duties of the Trustees (including, without limitation, the power of delegation) to any one or more Trustees, agents, representatives, officers, employees, independent contractors or other persons (including, but not limited to, the Manager) without regard to whether such power, authority or duty is normally granted or delegated by Trustees:
- (y) to collect, sue for and receive sums of money coming due to the Trust, and to engage in, intervene in, prosecute, join, defend, compromise, abandon or adjust, by

arbitration or otherwise, any actions, suits, proceedings, disputes, claims, demands or other litigation relating to the Trust, the assets of the Trust or the Trust's affairs, to enter into agreements therefore whether or not any suit is commenced or claim accrued or asserted and, in advance of any controversy, to enter into agreements regarding the arbitration, adjudication or settlement thereof;

- (z) to renew, modify, release, compromise, extend, consolidate or cancel, in whole or in part, any obligation to or of the Trust;
- (aa) to cause legal title to any of the assets of the Trust to be held by and/or in the name of the Trustees, or by and/or in the name of the Trust or one or more of the Trustees or any other person, on such terms, in such manner with such powers in such person as the Trustees may determine and with or without disclosure that the Trust or Trustees are interested therein; provided, however, that should legal title to any of the assets of the Trust be held by and/or in the name of any person or persons other than the Trust, the Trustees shall require such person or persons to execute a declaration of trust acknowledging that legal title to such assets is held in trust for the benefit of the Trust;
- (bb) to determine conclusively the allocation to capital, income or other appropriate accounts for all receipts, expenses, disbursements and property of the Trust;
- (cc) to prepare, sign and file or cause to be prepared, signed and filed any prospectus, offering memorandum or similar document, and any amendment thereto and all agreements contemplated therein or ancillary thereto or relating to or resulting from any offering of the Units or other securities issued or held by the Trust and to pay the cost thereof and related thereto out of the property of the Trust whether or not such offering is or was of direct benefit to the Trust or those persons (if any) who were Unitholders immediately prior to such offering;
- (dd) to make or cause to be made application for the listing on any stock exchange of any Units or other securities of the Trust, and to do all things which in the opinion of the Trustees may be necessary or desirable to effect or maintain such listing or listings;
- (ee) to issue Units and other securities of the Trust for such consideration as the Trustees may deem appropriate in their sole discretion, such issuance to be subject to the terms and conditions of this Declaration of Trust;
- (ff) to implement an internal reorganization of the assets of the Trust and/or any of the Trust's subsidiaries (including, without limitation, forming additional trusts or limited partnerships to be subsidiaries of the Trust);
- (gg) to enter into arrangements regarding the development of raw land and/or construction of real estate assets with a third party (including, without limitation, arrangements with the Manager or its affiliates);
- (hh) in addition to the mandatory indemnification provided for in Section 10.9, to the extent permitted by law to indemnify, or enter into agreements with respect to the

indemnification of any person with whom the Trust has dealings, including, without limitation, the Trustees, the Manager, the Transfer Agent or any escrow agent, to such extent as the Trustees shall determine;

- (ii) to determine conclusively the value of any or all of the property of the Trust from time to time and, in determining such value, to consider such information and advice as the Trustees, in their sole judgment, may deem material and reliable;
- (jj) to do all such acts and things and to exercise such powers as may be delegated to the Trustees by any person who co-owns real property with the Trust;
- (kk) to exercise from time to time any and all rights, powers and privileges in relation to all matters relating to the maximization of Unitholder value in the context of a response to an offer for Units or for all or substantially all of the property and assets of the Trust or any subsidiary of the Trust pursuant to an Offer (as defined in Section 14.6) including (i) any Unitholder rights protection plan either prior to or during the course of any Offer; (ii) any defensive action either prior to or during the course of any Offer; (iii) the preparation of any "Directors Circular" in response to any Offer; (iv) consideration on behalf of Unitholders and recommendations to Unitholders in response to any Offer; (v) any regulatory or court action in respect of any related matters and (vi) the carriage or all related and ancillary matters; and
- (ll) to do all such other acts and things as are incidental to the foregoing, and to exercise all powers that are necessary or useful to promote any of the purposes for which the Trust is formed and to carry out the provisions of this Declaration of Trust.

10.3 Further Powers of the Trustees

Subject to the provisions hereof, the Trustees shall have the power to prescribe any form provided for or contemplated by this Declaration of Trust and the Trustees may make, adopt, amend or repeal regulations (the "Trustees' Regulations") containing provisions relating to the conduct of the affairs of the Trust, the rights or powers of the Trustees and the rights or powers of the Unitholders, Special Voting Unitholders or officers of the Trust, provided that such regulations shall not be inconsistent with law or with this Declaration of Trust and shall not, in the opinion of the Trustees, be prejudicial to the Unitholders or Special Voting Unitholders. The Trustees shall also be entitled to make any reasonable decisions, designations or determinations not contrary to this Declaration of Trust which they may determine are necessary or desirable in interpreting, applying or administering this Declaration of Trust or in administering, managing or operating the Trust or the Trust Assets. To the extent of any inconsistency between this Declaration of Trust and any regulation, decision, designation or determination made by the Trustees, this Declaration of Trust shall prevail and such regulation, decision, designation or determination shall be deemed to be modified to eliminate such inconsistency. Any Trustees' Regulations, decisions, designations or determinations made pursuant to this Section shall be conclusive and binding upon all persons affected thereby. The Trustees shall also have such additional powers as may be approved by an Ordinary Resolution.

10.4 Banking

The banking activities of the Trust, or any part thereof (including, but without restricting the generality of the foregoing: the operation of the Trust's accounts; the making, signing, drawing, accepting, endorsing, negotiating, lodging, depositing or transferring of any cheques, promissory notes, drafts, acceptances, bills of exchange and orders for the payment of money; the giving of receipts for orders relating to any property of the Trust; the execution of any agreement relating to any such banking activities and defining the rights and powers of the parties thereto; and the authorizing of any officer of such banker to do any act or thing on the Trust's behalf to facilitate such banking activities) will be transacted with such banks, trust companies, or other firms or corporations carrying on a banking or similar business as the Trustees may designate, appoint or authorize from time to time, and will be transacted on the Trust's behalf by one or more officers of the Trust, or the Manager as the Trustees may designate, appoint or authorize from time to time.

10.5 Standard of Care and Duties

The Trustees, in exercising the powers and authority conferred upon them hereunder, shall act honestly and in good faith with a view to the best interests of the Trust and shall exercise that degree of care, diligence and skill that a reasonably prudent person would exercise in comparable circumstances. A Trustee will not be liable in carrying out his or her duties under this Declaration of Trust except in cases where the Trustee fails to act honestly and in good faith with a view to the best interests of the Trust or to exercise the degree of care, diligence and skill that a reasonably prudent person would exercise in comparable circumstances. The duties and standard of care of the Trustees provided as aforesaid are intended to be similar to, and not to be any greater than, those imposed on a director of a corporation governed by the *Business Corporations Act* (Ontario). Unless otherwise required by law, the Trustees will not be required to give surety bond or security in any jurisdiction for the performance of any duties or obligations hereunder. The Trustees will not be required to devote their entire time to the investments, business or affairs of the Trust.

10.6 Fees and Expenses

As part of the expenses of the Trust, the Trustees may pay or cause to be paid out of the Trust Assets, reasonable fees, costs and expenses incurred in connection with the administration and management of the Trust, including, without limitation, fees of auditors, accountants, lawyers, appraisers and other agents, consultants and professional advisors employed by or on behalf of the Trust, and the cost of reporting or giving notices to Voting Unitholders. All costs, charges and expenses properly incurred by the Trustees on behalf of the Trust will be payable out of the Trust Assets.

10.7 Determinations of Trustees Binding

All determinations of the Trustees that are made in good faith with respect to any matters relating to the Trust, including whether any particular investment or disposition meets the requirements of this Declaration of Trust, shall be final and conclusive and shall be binding upon the Trust and all Unitholders.

10.8 Limitations on Liability

- (1)None of the Trustees nor any officers of the Trust or trustees, officers or directors (if any) of the Trust's subsidiaries will be liable to the Trust or any Unitholder, Special Voting Unitholder or former Unitholder or Special Voting Unitholder (in each case whether registered or beneficial) for: (a) any action taken in good faith in reliance on any documents that are, prima facie, properly executed; (b) for any depreciation of, or loss to, the Trust incurred by reason of the sale of any Trust Assets; (c) for the loss or disposition of monies or securities; (d) for any other action or failure to act including, without limitation, the failure to compel in any way any former or acting Trustee to redress any breach of trust or for any failure by the Trust's subsidiaries to perform obligations or pay monies owed to the Trust; except for a breach of the obligations in Section 10.5. If the Trustees have retained an appropriate expert or advisor or other person whose profession lends credibility to a statement made by the professional person with respect to any matter connected with their duties under this Declaration of Trust, the Trustees may in good faith act or refuse to act based on the advice of such expert or advisor or professional person and, notwithstanding any provision of this Declaration of Trust, including, without limitation, the duty of care, diligence and skill set out in Section 10.5, the Trustees will not be liable for any action or refusal to act in good faith based on the advice of any such expert or advisor or professional person where it is reasonable to conclude that such advice is within the expertise of such expert or advisor or professional person to give.
- (2) The Trustees and officers of the Trust in incurring any debts, liabilities or obligations, or taking or omitting any other actions for or in connection with the affairs of the Trust are, and will conclusively be deemed to be, acting for and on behalf of the Trust, and not in their own personal capacities. None of the Trustees nor any officer of the Trust will be subject to any personal liability for any debts, liabilities, obligations, claims, demands, judgments, costs, charges or expenses (including legal expenses) against or with respect to the Trust or in respect to the affairs of the Trust. Other than the Trust Assets, no property or assets of the Trustees, owned in their personal capacity or otherwise, will be subject to any levy, execution or other enforcement procedure with regard to any obligations under this Declaration of Trust or under any other related agreements and no recourse may be had or taken, directly or indirectly, against the Trustees in their personal capacity or against any successor, heir, executor, administrator or legal representative of the Trustees. The Trust will be solely liable therefor and resort will be had solely to the Trust Assets for payment or performance thereof.

10.9 Indemnification

(1) Subject to Section 10.9(2), in addition to and without limiting any other protection hereunder or otherwise provided by law of each person who is, or shall have been, a Trustee, an officer of the Trust, or a trustee, officer or director (if any) of the Trust's subsidiaries or an officer or director of the Manager (and their respective heirs and legal representatives) or any other person acting in a similar capacity (regardless of whether that person is employed by a subsidiary, the Manager or any other person) (collectively, the "Indemnified Persons"), the Trust hereby agrees to indemnify, defend and save harmless the Indemnified Persons out of the Trust Assets, from and against any and all liabilities, losses, claims, damages, penalties, fines, actions, suits, demands, levies, assessments, costs, charges, expenses and disbursements (whether or not involving a third party claim), including legal expenses and including amounts paid to settle an action or satisfy a judgment (collectively, "Damages"), which may at any time be suffered by, imposed upon, incurred by or asserted against any of the Indemnified Persons (i) in respect of any civil, criminal or

administrative action or proceeding to which the Indemnified Person is made a party by reason of being or having been a Trustee or officer of the Trust or, at the request of the Trust, a trustee, officer or director (if any) of its subsidiaries and/or (ii) in respect of any and all taxes, penalties or interest in respect of unpaid taxes or other governmental charges imposed upon the Indemnified Person in consequence of his or her performance of his or her duties hereunder. An Indemnified Person will not be entitled to satisfy any right of indemnity or reimbursement granted herein, or otherwise existing under law, except out of the Trust Assets, and no Unitholder, Special Voting Unitholder or other Trustee or officer will be personally liable to any person with respect to any claim for such indemnity or reimbursement.

(2) An Indemnified Person shall not be indemnified under Section 10.9(1) in respect of Damages, or unpaid taxes or other governmental charges that (i) arise out of or as a result or in the course of his or her failure to act honestly and in good faith with a view to the best interests of the Trust or its subsidiaries, or (ii) in the case of a criminal or administrative action or proceeding that is enforced by a monetary penalty, where the Indemnified Person did not have reasonable grounds for believing that his or her conduct was lawful.

10.10 Contractual Obligations of the Trust

The Trustees shall use reasonable means where practicable to inform all persons having dealings with the Trust of the limitation of liability set forth in Section 10.8 and Section 2.8, and shall use reasonable means where practicable to cause to be inserted in any written agreement, undertaking or obligation made or issued on behalf of the Trust an appropriate statement of the disavowal and limitation of liability as set forth in Section 10.8 and Section 2.8, substantially in the following form:

"Each of the parties hereto acknowledges that the Trustees of the Trust are entering into this agreement solely in their capacity as Trustees of the Trust, and that the obligations or liabilities (including those arising hereunder or arising in connection herewith or from the matters to which this agreement relates, if any, including without limitation, claims based on negligence or otherwise tortious behavior) of the trustees, managers, officers, consultants, agents or employees of the Trust hereunder will not be binding upon, nor will resort be had to the property of, any of the holders of units of the Trust or any annuitant under a plan of which a holder of units is a trustee or carrier (an "annuitant"). The obligations or liabilities, if any, of the trustees, managers, officers or employees of the Trust hereunder shall be satisfied only out of the property of the Trust and no resort may be had to the property of any trustee, manager, officer or employee of the Trust. The provisions of this paragraph shall enure to the benefit of the heirs, successors, assigns and personal representatives of the trustees, managers, officers or employees of the Trust and of the holders of units and annuitants and, to the extent necessary to provide effective enforcement of such provisions, the trustees of the Trust are hereby acknowledged to be acting, and shall be entitled to act as, trustees for the holders of units and annuitants."

The omission of such statement from any such instrument will not render any Trustee, any Unitholder, Special Voting Unitholder or manager, officer, consultant, agent or employee of the Trust liable to any person, nor will any Trustee or any Unitholder, Special Voting Unitholder or any

manager, officer, consultant, agent or employee of the Trust be liable to any person for such omission. If, notwithstanding this provision, any Trustee, Unitholder, Special Voting Unitholder or any manager, officer, consultant, agent or employee of the Trust is held liable to any other person by reason of the omission of such statement from any such agreement, undertaking or obligation, such Trustee, Unitholder, Special Voting Unitholder or manager, officer, consultant, agent or employee will be entitled to indemnity out of the Trust Assets to the full extent of such liability and the costs of any litigation or other proceedings in which such liability has been determined, including without limitation, the fees and disbursements of counsel.

10.11 Conflicts of Interest

If a Trustee or an officer of the Trust is a party to a material contract or transaction or proposed material contract or transaction with the Trust, or is a director or officer or employee of, or has a material interest in, any person who is a party to a material contract or transaction or proposed material contract or transaction with the Trust, such Trustee or officer of the Trust, as the case may be, shall disclose in writing to the Trustees or request to have entered in the minutes of meetings of Trustees the nature and extent of such interest.

- (1) The disclosure required in the case of a Trustee or officer shall be made:
- (a) at the meeting of Trustees, as the case may be, at which a proposed contract or transaction is first considered;
- (b) if the Trustee or officer was not then interested in a proposed contract or transaction, at the first such meeting after he or she becomes so interested;
- (c) if the Trustee or officer becomes interested after a contract is made or a transaction is entered into, at the first meeting after he or she becomes so interested; or
- (d) if a person who is interested in a contract or transaction later becomes a Trustee or officer, at the first such meeting of Trustees after he or she assumes that capacity.
- (2) Notwithstanding paragraph (1), where this Section applies to any person in respect of a material contract or transaction or proposed material contract or transaction that, in the ordinary course of the business of the Trust, would not require approval by the Trustees or Voting Unitholders, such person shall disclose in writing to the Trustees or request to have entered in the minutes of meetings of Trustees the nature and extent of his or her interest forthwith after that person becomes aware of the contract or transaction or proposed contract or transaction.
- (3) A Trustee referred to in this Section shall not vote on any resolution to approve the contract or transaction unless the contract or transaction is:
 - (a) one relating primarily to his or her remuneration as a Trustee, officer, employee or agent of the Trust or its subsidiaries; or
 - (b) one for indemnity under Section 10.9 or for the purchase of liability insurance.
- (4) For the purposes hereof, a general notice to the Trustees by a Trustee or an officer of the Trust or any other person referred to in this Section disclosing that he or she is a director, officer

or employee of or has a material interest in a person and is to be regarded as interested in any contract made or any transaction entered into with that person, is a sufficient disclosure of interest in relation to any contract so made or transaction so entered into.

- (5) Where a material contract is made or a material transaction is entered into between the Trust and any one or more of its Trustees or officers, or between the Trust and another person of which a Trustee or officer of the Trust is a director or officer or in which he or she has a material interest:
 - (a) the Trustee or officer, as applicable, is not accountable to the Trust or to the Unitholders for any profit or gain realized from the contract or transaction; and
 - (b) the contract or transaction is neither void nor voidable;

by reason only of that relationship or by reason only that the Trustee is present at or is counted to determine the presence of a quorum at the meeting of Trustees or committee of Trustees that authorized the contract or transaction, if the Trustee disclosed his interest in accordance with this Section and the contract or transaction was reasonable and fair to the Trust at the time it was so approved.

- (6) Notwithstanding anything in this Section, but without limiting the effect of paragraph (5), a Trustee or officer of the Trust, acting honestly and in good faith, is not accountable to the Trust or to the Unitholders for any profit or gain realized from any such contract or transaction by reason only of his holding the office of Trustee or officer, and the contract or transaction, if it was reasonable and fair to the Trust at the time it was approved, is not by reason only of the Trustee's or officer's interest therein void or voidable, where:
 - (a) the contract or transaction is confirmed or approved at a meeting of Voting Unitholders duly called for that purpose; and
 - (b) the nature and extent of the Trustee's or officer's interest in the contract or transaction are disclosed in reasonable detail in the notice calling the meeting or in any information circular required to be provided by the Declaration of Trust or by law.
- (7) Subject to paragraphs (5) and (6) of this Section, where any Trustee or officer of the Trust fails to disclose his interest in a material contract or transaction in accordance with this Declaration of Trust or otherwise fails to comply with this Section, the Trustees or any Unitholder, in addition to exercising any other rights or remedies in connection with such failure exercisable at law or in equity, may apply to a court for an order setting aside the contract or transaction and directing that the Trustee or officer account to the Trust for any profit or gain realized.

10.12 Conditions Precedent

The obligation of the Trustees to commence or continue any act, action, suit or proceeding or to represent the Trust in any action, suit or proceeding shall be conditional upon sufficient funds being available to the Trustees from the Trust's property to commence or continue such act, action, suit or proceeding or to represent the Trust in any action, suit or proceeding and an indemnity reasonably satisfactory to the Trustees to protect and hold harmless the Trustees against the costs,

charges and expenses and liabilities to be incurred therein and any loss and damage they may suffer by reason thereof. None of the provisions contained in this Declaration of Trust shall require the Trustees to expend or risk their own funds or otherwise incur financial liability in the performance of their duties or in the exercise of any of their rights or powers unless they are given an indemnity and funding satisfactory to the Trustees, acting reasonably.

10.13 Execution of Instruments and Apparent Authority

- (1) The Trustees will have the power, from time to time, to appoint any Trustee or Trustees, any officer or officers of the Trust or any other person or persons on behalf of the Trust (including, without limitation, the Manager) either to sign agreements, instruments or other documents in writing and to do such acts and agree to such things, on behalf of the Trust, as may be necessary or desirable in connection therewith.
- Assets and any right, title or interest therein, or to the Trust or to the Units or Special Voting Units, will be entitled to rely on a certificate, statutory declaration or resolution executed or certified by a Trustee as to the capacity, power and authority of the Trustees, an officer, employee, consultant or agent of the Trust or any other person to act for and on behalf of and in the name of the Trust. No person dealing with a Trustee or any officer, employee, consultant or agent of the Trust will be bound to see to the application of any funds or property passing into the hands or control of such Trustee, officer, employee, consultant or agent of the Trust. The receipt of a Trustee or of authorized officers, employees, consultants or agents of the Trust, for moneys or other consideration, will be binding upon the Trust.

10.14 Declarations as to Beneficial Owner

The Trustees may require any Unitholder or holder of Special Voting Units as shown on the register of Unitholders or Special Voting Unitholder, as applicable, to provide a declaration, in form prescribed by the Trustees, as to the beneficial owner of Units or Exchangeable Securities registered in such Unitholder's or Special Voting Unitholder's name, as applicable, and as to the jurisdiction in which such beneficial owner is resident.

ARTICLE 11 AMENDMENT

11.1 Amendment

Subject to Section 11.2, the provisions of this Declaration of Trust, except where specifically provided otherwise, may only be amended by an Ordinary Resolution at a meeting of Voting Unitholders duly called for that purpose; provided that the provisions of this Declaration of Trust may be amended by the Trustees without the consent, approval or ratification of the Unitholders or any other person:

- (a) at any time and in any manner prior to the Closing; or
- (b) after the Closing for the purpose of:

- (i) ensuring continuing compliance with applicable laws (including the Tax Act and maintaining the Trust's status as a "mutual fund trust" under the Tax Act), regulations, requirements or policies of any governmental or other authority having jurisdiction over the Trustees, the Trust or over the distribution of Units and, to the extent reasonably practicable, ensuring that the Trust will not be a SIFT for the purposes of the SIFT Proposals or any final legislation implementing the SIFT Proposals;
- (ii) providing additional protection or added benefits, in the opinion of the Trustees, for the Voting Unitholders;
- (iii) removing any conflicts or inconsistencies in this Declaration of Trust or making corrections including the rectification of any ambiguities, defective provisions, errors, mistakes or omissions which are, in the opinion of the Trustees, necessary or desirable and not prejudicial to the Voting Unitholders;
- (iv) making amendments which, in the opinion of the Trustees, are necessary or desirable to remove conflicts or inconsistencies between the disclosure in the Information Circular and this Declaration of Trust;
- (v) making amendments of a minor or clerical nature or to correct typographical mistakes, ambiguities or manifest omissions or errors, which amendments, in the opinion of the Trustees, are necessary or desirable and not prejudicial to the Voting Unitholders;
- (vi) making amendments which, in the opinion of the Trustees, are necessary or desirable as a result of changes in GAAP (including accounting guidelines) or taxation or other laws or the administration or enforcement thereof; or
- (vii) making amendments which, in the opinion of the Trustees, are necessary or desirable to enable the Trust to issue Units for which the purchase price is payable on an installment basis; or
- (c) for any purpose (except one in respect of which the approval or consent of Voting Unitholders is specifically otherwise required) if the Trustees are of the opinion that the amendment is not prejudicial to Voting Unitholders and is necessary or desirable (including, without limitation, in connection with a reorganization contemplated by Section 11.4), but notwithstanding the foregoing, no such amendment shall modify the right to vote attached to any Unit or Special Voting Unit or the entitlement to distributions from the Trust provided hereunder (including those provided for in Article 5 and Article 14) represented by any Unit without the consent of the Unitholders pursuant to a Special Resolution and no such amendment shall reduce the percentage of votes required to be cast at a meeting of the Voting Unitholders for the purpose of this Section 11.1 without the consent of the Voting Unitholders pursuant to a Special Resolution or cause the Trust to fail or cease to qualify as a "mutual fund trust" under the Tax Act.

11.2 Special Resolution

Notwithstanding Section 11.1, at all times the following actions or amendments to the Declaration of Trust require approval by Special Resolution at a meeting of Unitholders or Voting Unitholders, as applicable, duly called for that purpose:

- (a) any combination, merger, amalgamation or arrangement of the Trust or any of the Trust's subsidiaries, as the case may be, any sale of all or substantially all of the assets of the Trust or any of the Trust's subsidiaries, as the case may be, or the liquidation or dissolution of the Trust or any of the Trust's subsidiaries, as the case may be, (other than in each case as part of an internal reorganization of the assets of the Trust and/or any of the Trust's subsidiaries, as the case may be, as approved by the Trustees);
- (b) any amendment to the provisions of Section 4.1 or to the provisions of Sections 4.2(b), provided that the remaining provisions of Section 4.2 may be amended by Ordinary Resolution;
- (c) an exchange, reclassification or cancellation of all or part of the Units or Special Voting Units;
- (d) the addition, change or removal of the rights, privileges, restrictions or conditions attached to the Units or Special Voting Units and, including, without limiting the generality of the foregoing:
 - (i) the removal or change of rights to distributions;
 - (ii) the addition or removal of or change to conversion privileges, options, voting, transfer or pre-emptive rights; or
 - (iii) the reduction or removal of a distribution preference or liquidation preference;
- (e) the termination of the Trust; or
- (f) the constraint on the issue, transfer or ownership of Units or Special Voting Units or the change or removal of such constraint, except as otherwise provided herein.

11.3 Variation of Rights

Except as otherwise provided herein, the rights and restrictions attached to the Units and Special Voting Units, respectively, may not be varied or abrogated without the consent of the holders of the outstanding Units or Special Voting Units, as the case may be, by Special Resolution voting separately as a class. There shall not otherwise be class votes of Units or Special Voting Units.

11.4 Internal Restructuring

Notwithstanding anything to the contrary herein contained, if at any time the Trustees so resolve to implement an internal reorganization of the assets of the Trust and/or any of the Trust's

subsidiaries (including, without limitation, forming additional trusts or limited partnerships to be subsidiaries of the Trust), any such resolution or reorganization shall not require the prior approval of Voting Unitholders.

11.5 No Termination

No amendment to or amendment and restatement of this Declaration of Trust, whether pursuant to this Article 11 or otherwise, shall be construed as a termination of the Trust or the settlement or establishment of a new trust.

ARTICLE 12 MEETINGS OF VOTING UNITHOLDERS

12.1 Annual and Special Meetings of Voting Unitholders

- (1) There shall be an annual meeting of the Voting Unitholders. Each annual meeting of the Voting Unitholders shall be called, commencing in 2008, on a day on or before June 30 in each year, at a time and at a place in Canada set by the Trustees. The business transacted at such meeting shall include:
 - (a) the presentation of the audited financial statements of the Trust for the immediately preceding fiscal year;
 - (b) the appointment of the Trustees for the ensuing year in accordance with Article 9;
 - (c) the appointment of Auditors; and
 - (d) the transaction of such other business as Voting Unitholders may be entitled to vote upon as herein provided or as may be properly brought before the meeting or as the Trustees may determine.
 - (2) Special meetings of the Voting Unitholders may be called at any time by the Trustees.
- (3) Voting Unitholders holding in the aggregate not less than 10% of all votes entitled to be voted at a meeting of Voting Unitholders may requisition the Trustees to call a special meeting of Voting Unitholders for the purposes stated in the requisition. The requisition shall (A) be in writing, (B) set forth the name and address of, and number of Units and Exchangeable Securities (and votes attached thereto which, in the aggregate, shall not be less than 10% of all votes entitled to be voted at a meeting of Unitholders) held by, each person who is supporting the requisition, and (C) shall state in reasonable detail the business to be transacted at the meeting and shall be sent to the Trustees. Upon receiving a requisition complying with the foregoing, the Trustees shall call a meeting of Unitholders to transact the business referred to in the requisition, unless:
 - (a) a record date for a meeting of Voting Unitholders has been fixed;
 - (b) the Trustees have called a meeting of Unitholders and have given notice thereof pursuant to Section 12.2; or
 - (c) in connection with the business as stated in the requisition:

- (i) the matter is not submitted to the Trust at least 60 days before the anniversary date of the last annual meeting, if the matter is proposed to be raised at an annual meeting, or at least 60 days before a special meeting other than an annual meeting, if the matter is proposed to be raised at a meeting other than the annual meeting;
- (ii) it appears that a matter covered by the requisition is submitted by the Voting Unitholder primarily for the purpose of enforcing a personal claim or redressing a personal grievance against the Trust or its subsidiaries, the Trustees, the Manager, the Voting Unitholders, officers or employees of the Trust and/or the Manager, or one or more of the Trust's subsidiaries, or primarily for the purpose of promoting general economic, political, religious, social or similar causes or primarily for a purpose that does not relate in a significant way to the business or affairs of the Trust;
- (iii) the Trust, at the request of one or more Voting Unitholders, had previously included a matter substantially the same as a matter covered by the requisition in an information circular relating to a meeting of Voting Unitholders held within 30 months preceding the receipt of such requisition and such Voting Unitholders failed to present the matter, in person or by proxy, at the meeting;
- (iv) substantially the same matter covered by the requisition was submitted to Voting Unitholders in an information circular relating to a meeting of Voting Unitholders held within 30 months preceding the receipt of such requisition and the matter covered by the requisition was defeated; or
- (v) the rights conferred by this Section 12.1(3) are being abused to secure publicity.
- (4) If the Trustees do not, within 21 days after receiving the requisition, call a meeting (except where the grounds for not calling the meeting are one or more of those set forth in subsection 12.1(3) above), any Unitholder who signed the requisition may call the meeting in accordance with the provisions of Article 12, *mutatis mutandis*.
- (5) The chair of any annual or special meeting shall be the Chair of the Trustees or any other Trustee specified by resolution of the Trustees or, in the absence of any Trustee, any person appointed as chair of the meeting by the Voting Unitholders present.
- (6) The Trustees, the officers of the Trust, the Auditors, Counsel and any other person(s), whose presence at the meeting is approved by (i) the Trustees, (ii) the chair of the meeting, or (iii) by Ordinary Resolution proposed at the meeting, may attend meetings of the Voting Unitholders.
- (7) Any person entitled to attend a meeting of Voting Unitholders may participate in the meeting, subject to and in accordance with applicable securities laws, if any, by means of a telephonic, electronic or other communication facility that permits all participants to communicate adequately with each other during the meeting, if the Trust makes available such a communication

facility. A person participating in a meeting by such means is deemed for the purposes of this Declaration of Trust to be present at the meeting.

(8) If the Trustees call a meeting of Voting Unitholders pursuant to this Declaration of Trust, the Trustees may determine that the meeting shall be held, subject to and in accordance with applicable securities laws, if any, entirely by means of a telephonic, electronic or other communication facility that permits all participants to communicate adequately with each other during the meeting.

12.2 Notice of Meetings

Notice of all meetings of Voting Unitholders will be given by unregistered mail, postage prepaid, addressed to each Voting Unitholder at his or her last address on the Register of the Trust, mailed at least 21 days and not more than 60 days before the meeting. Such notice will specify the time and place of such meeting and will specify the nature of the business to be transacted at such meeting in sufficient detail to permit a Voting Unitholder to form a reasoned judgment thereon, together with the text of any Special Resolution, at the time of mailing of the notice, proposed to be passed at such meeting. Any adjourned meeting may be held as adjourned without further notice. The accidental omission to give notice or the non-receipt of such notice by a Voting Unitholder will not invalidate any resolution passed at any such meeting. Notwithstanding the foregoing, a meeting of Voting Unitholders may be held at any time without notice if all the Voting Unitholders are present or represented thereat or those not so present or represented have waived notice. Any Voting Unitholder (or a duly appointed proxy of a Voting Unitholder) may waive any notice required to be given under the provisions of this Section 12.2, and such waiver, whether given before or after the meeting, will cure any default in the giving of such notice.

12.3 Quorum

At any meeting of the Voting Unitholders, a quorum will consist of two or more persons present in person either holding personally or representing as proxies in the aggregate at least 10% of the votes attached to all outstanding Voting Units. In the event of such quorum not being present at the appointed place on the date for which the meeting is called within 30 minutes after the time fixed for the holding of such meeting, the meeting, if called by request of Voting Unitholders other than pursuant to Section 9.8, will be terminated (and not adjourned) and, if otherwise called, will stand adjourned to a day not less than 7 days later and to such place and time as may be appointed by the chairperson of the meeting. If at such adjourned meeting a quorum as above defined is not present, the Voting Unitholders present either personally or by proxy will be deemed to form a quorum, and any business may be brought before or dealt with at such an adjourned meeting that might have been brought before or dealt with at the original meeting in accordance with the notice calling the same.

12.4 Voting Rights

Only Voting Unitholders of record will be entitled to vote. Each Unit will entitle the holder or holders of that Unit to one vote on a ballot vote at any meeting of Voting Unitholders. Each Special Voting Unit will entitle the holder or holders thereof to that number of votes on a ballot vote at any meeting of Voting Unitholders that is equal to the number of Units into which the Exchangeable Security to which such Special Voting Unit is attached is, directly or indirectly, exchangeable or convertible. Voting will, unless a ballot vote is demanded, be by a show of hands. A

Voting Unitholder or proxy holder may demand a ballot vote either before or after any vote by show of hands. At any meeting of Voting Unitholders, any holder of Voting Units entitled to vote at such meeting may vote by proxy and a proxy holder need not be a Voting Unitholder, provided that, unless otherwise determined by the Trustees, no proxy will be voted at any meeting unless it has been received by the Transfer Agent (or if none, the Trust) for verification at least 48 hours prior to the commencement of such meeting. When any Voting Unit is held jointly by several persons, any one of them may vote at any meeting in person or by proxy in respect of such Voting Unit, but if more than one of them is present at such meeting in person or by proxy they shall vote as one with respect to the Voting Units jointly held by them. Notwithstanding the foregoing, the Trustees may establish rules for voting at a meeting of Voting Unitholders to be carried out by means of telephone, electronic or other communication facility.

12.5 Resolutions Binding the Trustees

Voting Unitholders will be entitled to pass resolutions that will bind the Trust or the Trustees only with respect to the following matters:

- (a) the election or removal of Trustees;
- (b) the appointment or removal of Auditors;
- (c) amendments to the Declaration of Trust (as provided in Article 11);
- (d) the termination of the Trust as provided in Article 15;
- (e) the actions set out in Section 11.2; and
- (f) such other business as the Trustees may determine or as may properly be brought before the Voting Unitholders, including without limitation, any other matters required by securities law, stock exchange rules or other laws or regulations to be submitted to Voting Unitholders for their approval.

Except with respect to the matters set out in this Section 12.5, no action taken by the Voting Unitholders or any resolution of the Voting Unitholders at any meeting will in any way bind the Trust or the Trustees. Any action taken or resolution passed in respect of any matter at a meeting of Voting Unitholders will be by Special Resolution, except for the matters set out in subsections 12.5(a), 12.5(b) and 12.5(c) above, which matters may be dealt with by Ordinary Resolution unless the contrary is otherwise expressly provided under any specific provision of this Declaration of Trust.

12.6 Meaning of "Special Resolution"

(1) The expression "**Special Resolution**" when used in this Declaration of Trust means, subject to this Article 12, a resolution proposed to be passed as a special resolution at a meeting of Voting Unitholders (including an adjourned meeting) duly convened for that purpose and held in accordance with the provisions of this Section at which a quorum is present and passed by the affirmative votes of the holders of at least two-thirds of the Voting Units represented at the meeting and voted on a poll upon such resolution, or approved in writing by the holders of not less than two-thirds of the Voting Units entitled to vote on such resolution.

- (2) For the purpose of a separate class vote by the holders of Units or Special Voting Units as a class as provided herein, the expression "Special Resolution" means a resolution proposed to be passed at a separate meeting of holders of Units or Special Voting Units, as the case may be, at which meeting the provisions of this Article 12 shall apply *mutatis mutandis*.
- (3) Notwithstanding Section 12.3, if at any meeting at which a Special Resolution is proposed to be passed, the holders of 10% of the aggregate number of votes attached to Voting Units outstanding are not present in person or by proxy within 30 minutes after the time appointed for the meeting, then the meeting, if convened by or on the requisition of Voting Unitholders, shall be terminated (and not adjourned); but in any other case it shall stand adjourned to such date, being not less than 21 nor more than 60 days later and to such place and time as may be appointed by the chair of the meeting. Not less than 10 days prior notice shall be given of the time and place of such adjourned meeting in the manner provided in Section 12.2. Such notice shall state that at the adjourned meeting the Voting Unitholders present in person or by proxy shall form a quorum, but it shall not be necessary to set forth the purposes for which the meeting was originally called or any other particulars. At the adjourned meeting, the Voting Unitholders present in person or by proxy shall form a quorum and may transact the business for which the meeting was originally convened and a resolution proposed at such adjourned meeting and passed by the requisite vote as provided in Sections 12.6(1) or 12.6(2) shall be a Special Resolution within the meaning of this Declaration of Trust, notwithstanding that the holders of less than 10% of the aggregate number of votes attached to Voting Units then outstanding are present in person or by proxy at such adjourned meeting.

12.7 Meaning of "Outstanding"

Every Voting Unit issued, certified and delivered hereunder will be deemed to be outstanding until it is cancelled or delivered to the Trustees or Transfer Agent for cancellation, provided that:

- (a) when a new certificate has been issued in substitution for a Unit Certificate that has been lost, stolen, mutilated or destroyed, only the later of such Unit Certificates will be counted for the purposes of determining the number of Voting Units outstanding;
- (b) for the purpose of any provision of this Declaration of Trust entitling holders of outstanding Voting Units to vote, sign consents, requisitions or other instruments or take any action under this Declaration of Trust, Voting Units owned directly or indirectly, legally or equitably, by the Trust, or any subsidiary thereof will be disregarded, except that:
 - (i) for the purpose of determining whether the Trustees will be protected in relying on any such vote, consent, requisition or other instrument or action, only the Voting Units that the Trustees know are so owned will be so disregarded; and
 - (ii) Voting Units so owned that have been pledged in good faith other than to the Trust or a subsidiary thereof will not be so disregarded if the pledgee establishes to the satisfaction of the Trustees the pledgee's right to vote such Voting Units in his or her discretion free from the control of the Trust or any subsidiary thereof; and

(c) for the purposes of Section 12.7(b), any Trustee, any officer of the Trust or the Transfer Agent will provide a certificate that will state the number of Voting Units and the certificate numbers of certificates, if certificates are issued, held by the Trust or any subsidiary thereof. The Trustees will be entitled to rely on such certificate in order to disregard the votes of any of the parties mentioned above.

12.8 Record Date for Voting

For the purpose of determining the Voting Unitholders who are entitled to receive notice of and vote or act at any meeting or any adjournment thereof or for the purpose of any other action, the Trustees may fix a date as a record date for the determination of Voting Unitholders entitled to receive notice of and/or to vote at such meeting or any adjournment thereof or to be treated as Voting Unitholders of record for purposes of such other action, and any Voting Unitholder who was a Voting Unitholder at the time so fixed will be entitled to receive notice of and vote at such meeting or any adjournment thereof or to be treated as a Voting Unitholder for purposes of such other action even though the Voting Unitholder has since that time disposed of his or her Voting Units, and no Voting Unitholder becoming such after that time will be so entitled to vote at such meeting or any adjournment thereof or to be treated as Voting Unitholders for purposes of such other action. In the event that the Trustees do not fix a record date for any meeting of Voting Unitholders, the record date for receiving notice of and voting at such meeting will be the date upon which notice of the meeting is given as provided under Section 12.2. This Section 12.8 is subject to Section 6.2(3).

12.9 Proxies

- this Declaration of Trust, such vote or consent may be given either directly by the Voting Unitholder or by a proxy. The instrument appointing a proxy shall be in writing and either substantially in a form which may be approved by the Trustees acting reasonably or as may be satisfactory to the chair of the meeting at which it is sought to be exercised. The instrument of proxy shall be executed by the Voting Unitholder giving the proxy or his or her agent duly authorized in writing and, if given on behalf of joint holders, shall be executed by all of them and may be revoked by any of them, and, if given by a Voting Unitholder which is a body corporate, shall be executed on its behalf by a person duly authorized in writing. Any person may be appointed a proxy, whether or not that person is a Voting Unitholder. The Trustees, on behalf of the Trust, may solicit instruments of proxy from the Voting Unitholders or any of them in respect of any matter requiring or permitting the Voting Unitholders' vote or consent. An instrument of proxy shall be deposited with the chair of the meeting before any vote is cast under its authority or at such earlier time or in such manner as the Trustees may prescribe from time to time.
- (2) An instrument of proxy executed in compliance with the foregoing shall be valid unless challenged at the time of or prior to its exercise and the person challenging the instrument shall have the burden of proving, to the satisfaction of the chair of the meeting at which the instrument is proposed to be used, that the instrument of proxy is invalid. Any decision of the chair of the meeting in respect of the validity of an instrument of proxy shall be final and binding upon all persons. An instrument of proxy shall be valid only at the meeting with respect to which it was solicited or any adjournment thereof.

(3) A vote cast in accordance with any proxy shall be valid notwithstanding the death, incapacity, insolvency or bankruptcy of the Voting Unitholder giving the proxy or the revocation of the proxy unless written notice of the death, incapacity, insolvency or bankruptcy of the Voting Unitholder or revocation of the proxy has been received by the chair of the meeting prior to the time the vote is cast.

12.10 Personal Representatives

Subject to the provisions of Article 13, if a Voting Unitholder is deceased, his personal representative, upon filing with the secretary of the meeting such proof of his or her appointment as the secretary considers sufficient, shall be entitled to exercise the same voting rights at any meeting of Voting Unitholders as the Voting Unitholder would have been entitled to exercise if the Voting Unitholder was living and for the purpose of the meeting shall be considered to be a Voting Unitholder. Subject to the provisions of the will of a deceased Voting Unitholder, if there is more than one personal representative, the provisions of Section 12.4 relating to joint holders shall apply.

12.11 Conduct of Meetings

To the extent that the rules and procedures for the conduct of a meeting of Voting Unitholders are not prescribed herein, the rules and procedures shall be such reasonable rules and procedures as are determined by the chair of the meeting and such rules and procedures shall be binding upon all parties participating in the meeting.

12.12 Binding Effect of Resolutions

Every resolution passed at a meeting in accordance with the provisions of this Declaration of Trust shall be binding upon all Voting Unitholders, whether present at or absent from the meeting. Subject to Section 12.5, no action taken by Voting Unitholders at any meeting of Voting Unitholders shall in any way bind the Trust or the Trustees without the approval of the Trustees.

12.13 Resolution in Writing

Notwithstanding any other provision of this Declaration of Trust, a resolution in writing circulated to all Voting Unitholders not less than 10 days prior to its effective date and executed by Voting Unitholders holding more than 50% of the outstanding Voting Units entitled to be voted on such resolution, if such resolution is an Ordinary Resolution, or a resolution in writing circulated to all Voting Unitholders and executed by Voting Unitholders holding more than two-thirds of the outstanding Voting Units entitled to be voted on such resolution, if such resolution is a Special Resolution, will be as valid and binding for all purposes of this Declaration of Trust as if such Voting Unitholders had exercised at that time all of the voting rights to which they were then entitled under Section 12.4 or Section 12.6 in favour of such resolution at a meeting of Voting Unitholders duly called for the purpose.

ARTICLE 13 CERTIFICATES AND REGISTRATION OF UNITS AND SPECIAL VOTING UNITS

13.1 Nature and Evidence of Voting Units

- (1) The provisions of this Article 13 shall not in any way alter the nature of the Units or the Special Voting Units or the relationship of a Voting Unitholder to the Trustees and of one Voting Unitholder to another but are intended only to facilitate the issuance of certificates, if desirable to issue them to Unitholders, evidencing the ownership of Units and the recording of all transactions in respect of Units, Special Voting Units and Unit Certificates whether by the Trust, securities dealers, stock exchanges, transfer agents, registrars or other persons.
- Unit Certificate") may be issued in the name of and deposited by the Transfer Agent with, or on behalf of, CDS or a successor (collectively, the "Depository") as custodian of such Global Unit Certificate and registered by the Transfer Agent in the name of the Depository or its nominee. No beneficial holder of Units represented in part by a Global Unit Certificate will be entitled to a certificate or other instrument from the Trust or the Depository evidencing that beneficial holder's ownership thereof except in the circumstances where the Depository resigns or is removed from its responsibilities as depository and the Trust is unable or does not wish to locate a qualified successor. Beneficial interests in a Global Unit Certificate will be represented only through the Book-Entry System. Transfers of Units between CDS Participants shall occur in accordance with the Depository's rules and procedures.
- All references herein to actions by, notices given or payments made to Voting Unitholders shall, where such Units are held through the Depository, refer to actions taken by, or notices given or payments made to, the Depository upon instruction from the CDS Participants in accordance with the Depository's rules and procedures. For the purposes of any provision hereof requiring or permitting actions with the consent of or at the direction of Voting Unitholders evidencing a specified percentage of the aggregate Voting Units outstanding, such direction or consent may be given by Voting Unitholders acting through the Depository and the CDS Participants owning Units evidencing the requisite percentage of the Voting Units. The rights of a Voting Unitholder whose Voting Units are held through the Depository shall be exercised only through the Depository and the CDS Participants and shall be limited to those established by law and agreements between such Voting Unitholders and the Depository and/or the CDS Participants or upon instruction from the CDS Participants. Each of the Transfer Agent and the Trustees may deal with the Depository for all purposes (including the making of payments) as the authorized representative of the respective Unitholders and such dealing with the Depository shall constitute satisfaction or performance, as applicable, towards their respective obligations hereunder, subject to the voting rights of the holders of record of Special Voting Units.
- (4) If any notice or other communication is required to be given to holders of Voting Units held through the Depository, the Trustees and the Transfer Agent shall give all such notices and communications to the Depository. If any notice or other communication is required to be given to direct holders of Voting Units, the Trustees and the Transfer Agent shall deliver all such notices and communications to the registered owners of such Voting Units.

(5) If: (i) the Trust determines that the Depository is no longer willing or able to discharge properly its responsibilities as depository with respect to the Units and the Trust is unable to locate a qualified successor; (ii) the Trust at its option elects, or is required by law, to terminate the Book-Entry System with respect to the Units; or (iii) Voting Unitholders determine by Special Resolution that the continuation of the Book-Entry System with respect to the Units is no longer in the best interest of the Unitholders, then the Depository shall surrender the Global Unit Certificate to the Transfer Agent with instructions from the Depository for registration of Units in the name and in the amounts specified by the Depository and the Trust shall issue and the Trustees and Transfer Agent shall execute and deliver the aggregate number of Units then outstanding in the form of definitive Unit Certificates representing such Units.

13.2 Liability

The Trust, the Trustees and the Transfer Agent will not have any liability for:

- (a) records maintained by CDS relating to the beneficial interests in the Units or the book entry accounts maintained by CDS;
- (b) maintaining, supervising or reviewing any records relating to such beneficial interests; or
- (c) any advice or representation made or given by CDS with respect to the rules and regulations of CDS or any action taken or not taken by CDS or the CDS Participants.

13.3 Unit Certificates

- (1) Unit Certificates shall, subject to the provisions hereof, be in such form as may be authorized from time to time by the Trustees.
 - (2) If issued, Unit Certificates are issuable only in fully registered form.
 - (3) The definitive form of the Unit Certificates will:
 - (a) be in the English language;
 - (b) be dated as of the date of issue thereof,
 - (c) contain the CUSIP number (if any) for the Units; and
 - (d) contain such distinguishing letters and numbers as the Trustees shall prescribe.
- (4) In the event that the form of Unit Certificate is translated into the French language and any provision of Unit Certificates in the French language is susceptible to an interpretation different from the equivalent provision in the English language, the interpretation of such provision in the English language shall be determinative.
- (5) Each Unit Certificate shall be signed on behalf of the Trustees (including by their attorney appointed under Section 17.11) and the Transfer Agent of such Units. The signature on behalf of the Trustees required to appear on such certificate may be printed, lithographed or

otherwise mechanically reproduced thereon and, in such event, certificates so signed are as valid as if they had been signed manually.

13.4 Contents of Unit Certificates

- (1) Until otherwise determined by the Trustees, each Unit Certificate will legibly set forth on the face thereof, *inter alia*, the following:
 - (a) the name of the Trust and the words "A trust established under the laws of the Province of Ontario by a Declaration of Trust dated March 27, 2007, as further amended or amended and restated" or words of like effect;
 - (b) the name of the person to whom the Unit Certificate is issued as Unitholder;
 - (c) the number and class of Units represented thereby, and that the Units represented thereby are fully paid;
 - (d) that the Units represented thereby are transferable;
 - (e) the words "The Units represented by this certificate are issued upon the terms and subject to the conditions of the Declaration of Trust, which Declaration of Trust is binding upon all holders of Units and, by acceptance of this certificate, the holder assents to the terms and conditions of the Declaration of Trust. A copy of the Declaration of Trust pursuant to which this certificate and the Units represented hereby are issued may be obtained by a Unitholder on demand and without fee from the head office of the Trust" or words of like effect.
 - (f) "For information as to personal liability of a Unitholder, see the reverse side of this certificate" or words of like effect.
- (2) Until otherwise determined by the Trustees, each such certificate will legibly set forth on the face or the reverse side thereof, *inter alia*, the following:
 - (a) "The Declaration of Trust provides that no Unitholder will be subject to any personal liability whatsoever, in tort, contract or otherwise, to any person in connection with the assets of the Trust or the obligations or the affairs of the Trust, and all such persons shall look solely to the assets of the Trust for satisfaction of claims of any nature arising out of or in connection therewith, and the assets of the Trust only shall be subject to levy or execution", or words of like effect; and
 - (b) appropriate forms of notice of exercise of the right of redemption and of powers of attorney for transferring Units (which may request the jurisdiction of residence of the beneficial transferees).
- (3) The Unit Certificates may be engraved, printed or lithographed, or partly in one form and partly in another, as the Trustees may determine.

13.5 Register of Unitholders and Special Voting Unitholders

A register of holders of Units and Special Voting Units (the "Register") will be kept on behalf of the Trust and will contain the names and addresses of the holders of Units and Special Voting Units, the respective numbers and classes of Units and Special Voting Units held by them, the certificate numbers of certificates, if any, representing such Units and Special Voting Units, and a record of all transfers and redemptions thereof. Registers will be maintained at such office or offices of the Transfer Agent as the Trustees may from time to time designate. Only Unitholders and Special Voting Unitholders whose Units and Special Voting Units are so recorded on the Register will be entitled to receive distributions, if applicable, or to exercise or enjoy the rights of Unitholders and Special Voting Unitholders hereunder. The Trustees will have the right to treat the person registered as a Unitholder or Special Voting Unitholder on the Register as the owner of such Units or Special Voting Units for all purposes, including, without limitation, payment of any distribution, if applicable, giving notice to Unitholders and Special Voting Unitholders and determining the right to attend and vote at meetings of Voting Unitholders. The Trustees will not be bound to recognize any transfer or attempted transfer, pledge or other disposition of a Unit or Special Voting Unit, or any equitable or other claim with respect thereto, whether or not the Trustees have actual or other notice thereof, until such Unit(s) or Special Voting Units have been transferred on the Register as herein provided. Notwithstanding the foregoing, in the event Units are issued in the Book-Entry System, the provisions of Section 13.1(2) and Section 13.1(3) shall apply in respect of such Units.

13.6 Transfer Agents and Registrars

The Trustees will enter into a registrar and transfer agency agreement with the Transfer Agent according to which the Transfer Agent will agree to act as transfer agent and registrar for the Units and the Special Voting Units, and the Trustees may provide for the transfer of Units and Special Voting Units in one or more places within or outside Canada (provided that if such appointments are made, there shall be a transfer agent and registrar within the Province of Ontario). Such Transfer Agent will perform those functions and duties usually performed by transfer agents and registrars of shares of corporations having share capital, including maintaining the Register and all other necessary or appropriate books and records (which may be kept on a computer or similar device) for recording original issues and registering and transferring and redeeming the Units and the Special Voting Units, and will to the extent practicable monitor the jurisdiction of residence of the beneficial owners of Units and Special Voting Units as set out in Section 14.1. In the event that the agreement with the Transfer Agent is terminated in accordance with its terms, the Trustee may appoint one or more chartered banks or banking institutions, trust companies or other persons to act as registrar and transfer agent upon terms satisfactory to the Trustees. In the case of an original issue of Units or Special Voting Units, the Transfer Agent may rely and act upon the written instruction of the Trustees without inquiry into the receipt by the Trust of, or the sufficiency of, the consideration for such original issue. The Trustees will cause similar registers to be created and maintained by or on behalf of the Trust for other securities issued by the Trust, where appropriate. The Trustees will have no liability for any actions of the Transfer Agent and the Trustees, in relying in good faith upon the Transfer Agent, will be deemed to have complied with their obligations under Section 10.5.

13.7 Lost Certificates

In the event that any certificate for Units is lost, stolen, destroyed or mutilated, the Trustees may authorize the issuance of a new certificate for the same number of Units in lieu thereof. The

Trustees may in their sole discretion, before the issuance of such new certificate, require the Unitholder of the lost, stolen, destroyed or mutilated certificate, or the legal representative of the Unitholder, to make an affidavit or statutory declaration setting forth such facts as to the loss, theft, destruction or mutilation as the Trustees may deem necessary, to surrender any mutilated certificate and shall require the applicant to supply to the Trust a "lost certificate bond" or a similar bond in such reasonable sum as the Trustees or the Transfer Agent may direct indemnifying the Trust for so doing.

ARTICLE 14 OWNERSHIP AND TRANSFER OF UNITS AND SPECIAL VOTING UNITS

14.1 Limitation of Non-Resident Ownership

- (1) It is hereby acknowledged that it is in the best interest of Unitholders that the Trust always qualify as a "mutual fund trust" under the Tax Act. Accordingly:
 - (a) in order to ensure that the Trust continues to qualify as a "mutual fund trust":
 - (i) the Trust shall not be maintained primarily for the benefit of Non residents and it shall be the responsibility of the Trustees to monitor the holdings by Non residents;
 - (ii) notwithstanding any delegation of the powers of the Trustees to another person, the Trustees shall and shall be entitled to take all such steps as are necessary or desirable to ensure that the Trust is not maintained primarily for the benefit of Non residents;
 - (b) the Trustees may, at any time and from time to time, make reasonable efforts, as practicable in the circumstances, to obtain declarations as to beneficial ownership under Section 10.14, perform residency searches of holders of securities of the Trust and beneficial holder mailing address lists and take such other steps specified by the Trustees, at the cost of the Trust, to determine or estimate in the circumstances the residence of the beneficial owners of securities of the Trust;
 - (c) if at any time the Trustees, in their sole discretion, determine that it is in the best interest of the Trust, the Trustees, may:
 - (i) require the Trust to refuse to accept a subscription for securities of the Trust from, or issue or register a transfer of securities of the Trust to, a person unless the person provides a declaration to the Trustees pursuant to Section 10.14 that the securities of the Trust to be issued or transferred to such person will not when issued or transferred be beneficially owned by a Non-resident;
 - (ii) to the extent practicable in the circumstances, send a notice to registered holders of securities of the Trust which are beneficially owned by Nonresidents, chosen in inverse order to the order of acquisition or registration of such securities beneficially owned by Non-residents or in such other manner as the Trustees may consider equitable and practicable, requiring them to sell

their securities of the Trust which are beneficially owned by Non-residents or a specified portion thereof within a specified period of not less than 60 days or such shorter period as may be required to preserve the status of the Trust as a mutual fund trust under the Tax Act. The Trustees may also refuse the issuance of Units to facilitate an exchange of Exchangeable Securities if to do so would cause the Trust to lose its status as a mutual fund trust under the Tax Act. If the holders of securities of the Trust receiving such notice have not sold the specified number of such securities or provided the Trustees with satisfactory evidence that such securities are not beneficially owned by Nonresidents within such period, the Trustees may, on behalf of such registered Unitholder, sell such securities and, in the interim and to the extent applicable, suspend the voting and distribution rights attached to such securities of the Trust and make any distribution in respect of such securities by depositing such amount in a separate bank account in a Canadian chartered bank (net of any applicable taxes). Any sale shall be made on any stock exchange on which the applicable securities of the Trust are then listed or in such other manner in which the Trustees shall determine, and upon such sale, the affected holders shall cease to be holders of such securities so disposed of and their rights shall be limited to receiving the net proceeds of sale, and any distribution in respect thereof deposited as aforesaid, net of applicable taxes and costs of sale, upon surrender of the certificates representing such securities;

- (iii) delist any listed securities of the Trust from any non-Canadian stock exchange; and/or
- (iv) take such other actions as the Trustees determine, in their sole discretion, are appropriate in the circumstances that will reduce or limit the number of securities of the Trust held by Non-residents to ensure that the Trust is not maintained primarily for the benefit of Non-residents.
- (2) The Trustees shall have no liability for amounts received pursuant to sales of securities of the Trust made pursuant to Section 14.1(1)(c)(ii). Except as specifically set out herein, the Trustees shall not be bound to do or take any proceeding or action with respect to this Section 14.1 by virtue of the powers conferred on them under this Declaration of Trust. It is acknowledged that the ability of the Trustees to monitor the Non resident holders of the Units will be limited due to the fact that the Units will be registered in the name of depositories and other non-beneficial holders and, in this regard, the Trustees shall be entitled to rely on information respecting the residency of Unitholders provided to the Trustees by the Transfer Agent and CDS Participants. The Trustees shall not be liable for any violation of the Non-resident ownership restriction in this Section 14.1 which may occur during the term of the Trust.
- (3) Notwithstanding any other provision of the Declaration of Trust, Non-residents, whether registered holders or beneficial holders of securities of the Trust, shall not be entitled to vote in respect of any Special Resolutions to amend this Section 14.1.

14.2 Transfer of Units

- (1) Units will be, for all purposes of the Trust and this Declaration of Trust, personal and moveable property, and will, subject to Section 14.1, be transferable at any time and from time to time. Transfers of Units will be recorded on the Register and will only become effective when so recorded. If Units are issued in the Book-Entry System, the provisions of Sections 13.1(2) and 13.1(3) will apply.
- (2) Special Voting Units will not be transferable without the contemporaneous transfer of the Exchangeable Securities to which such Special Voting Units are attached, and then only if the Transfer Agent has been furnished with evidence acceptable to it, in its sole discretion, of the transfer of the Exchangeable Securities to which such Special Voting Units are attached and evidence that the prospective transferee is not a Non-resident. Subject to the foregoing, transfers of Special Voting Units will be recorded on the Register and will only become effective when so recorded. Notwithstanding the foregoing, in the event of an Offer (as defined in Section 14.6(1)) by any person to purchase Exchangeable Securities that is made on identical terms to all Unitholders, the Special Voting Units to which such Exchangeable Securities are attached may be transferred by the registered holder thereof pursuant to such Offer without regard to the provisions of this Section 14.2(2).
- (3) No transfer of a Unit or Special Voting Unit will be recognized unless such transfer is of a whole Unit or Special Voting Unit, unless otherwise determined by the Trustees.
- (4) Subject to the provisions of this Article 14, Units and Special Voting Units will be transferable on the Register only by the Unitholders or Special Voting Unitholders of record thereof or their executors, administrators or other legal representatives or by their agents or attorneys duly authorized in writing, and, with respect to transfers of Units, only upon delivery to the Trust or to the Transfer Agent of the Unit Certificate therefor, properly endorsed or accompanied by a duly executed instrument of transfer or power of attorney and accompanied by all necessary transfer or other taxes imposed by law, together with such evidence of the genuineness of such endorsement, execution and authorization and other matters that may reasonably be required by the Trustees or the Transfer Agent. Upon such delivery the transfer will be recorded on the Register and a new Unit Certificate for the Units transferred will be issued to the transferee and a new Unit Certificate for the balance of Units not transferred will be issued to the transferor.
- (5) Subject to the provisions of this Article 14, any person becoming entitled to any Units or Special Voting Units as a consequence of the death, bankruptcy or incompetence of any Unitholder or Special Voting Unitholder, or otherwise by operation of law, will be recorded as the holder of such Units or Special Voting Units and, in the case of Units, will receive a new Unit Certificate therefor, only upon production of evidence satisfactory to the Trustees or the Transfer Agent and, in the case of Units, upon delivery of the existing Unit Certificate to the Trustees or the Transfer Agent, but until such record is made the Unitholder or Special Voting Unitholder of record will continue to be and be deemed to be the holder of such Units or Special Voting Units for all purposes whether or not the Trust, the Trustees or the Transfer Agent have actual or other notice of such death or other event.
- (6) Unit Certificates representing any number of Units may be exchanged without charge for Unit Certificates representing an equivalent number of Units in the aggregate. Any exchange of

Unit Certificates may be made at the offices of the Trust or the Transfer Agent where the Register is maintained pursuant to the provisions of this Article 14. Any Unit Certificates tendered for exchange shall be surrendered to the Trustees or the Transfer Agent and will then be cancelled.

14.3 Units and Special Voting Units Held Jointly or in a Fiduciary Capacity

Except as herein provided, the Trustees may treat two or more persons holding any Units or Special Voting Units as joint owners of the entire interest therein, and no entry will be made in the Register or on any certificate that any person is in any manner entitled to any future, limited or contingent interest in any Unit or Special Voting Unit; provided, however, that any person recorded as a Unitholder or Special Voting Unitholder may, subject to the provisions herein contained, be described in the Register or on any certificate as a fiduciary of any kind and any customary words may be added to the description of the holder to identify the nature of such fiduciary relationship, but as set forth in Section 14.4 the same shall not bind the Trust, the Trustees or the Transfer Agent.

14.4 Performance of Trust

The Trustees, the Trust and the Transfer Agent will not be bound to be responsible for or otherwise inquire into or ensure the performance of any trust, express, implied or constructive, or of any charge, pledge, security interest or equity to which any of the Units or Special Voting Units or any interests therein are or may be subject, or to ascertain or enquire whether any transfer of any such Units or Special Voting Units or interests therein by any such Unitholder or Special Voting Unitholder or by his or her personal representatives is authorized by such trust, charge, pledge, security interest or equity, or to recognize any person as having any interest therein except for the person recorded as the Unitholder or Special Voting Unitholder.

14.5 Death or Disability of a Unitholder or Special Voting Unitholder

The death or disability of a Unitholder or Special Voting Unitholder during the continuance of the Trust will not terminate the Trust or any of the mutual or respective rights and obligations created by or arising under this Declaration of Trust nor give such Unitholder's or Special Voting Unitholder's personal representatives a right to an accounting or to take any action in court or otherwise against other Unitholders or Special Voting Unitholder or the Trustees or the Trust Assets. The death of a Unitholder or Special Voting Unitholder will merely entitle the personal representatives of the deceased Unitholder or Special Voting Unitholder, as the case may be, to demand and receive, pursuant to the provisions hereof, a new certificate for Units or Exchangeable Securities in place of the certificate held by the deceased Unitholder or Special Voting Unitholder, as the case may be, if any, and upon the acceptance thereof such personal representatives will succeed to all rights of the deceased Unitholder or Special Voting Unitholder, as the case may be, under this Declaration of Trust.

14.6 Offer for Units

- (1) In this Section 14.6:
- (a) "**Dissenting Offeree**" means, where an Offer is made for all the Units, a holder of a Unit who does not accept the Offer, and includes a subsequent holder of a Unit that is

- the subject of the Offer, including a subsequent holder of Units who acquires such Units upon the conversion or exchange of Exchangeable Securities;
- (b) "Offer" means an offer to acquire, directly or indirectly, outstanding Units where, as of the date of the offer to acquire, the Units that are the subject of the offer to acquire, together with the Offeror's Units, constitute in the aggregate 20% or more of all outstanding Units;
- (c) "offer to acquire" includes an acceptance of an offer to sell;
- (d) "Offeror" means a person, or two or more persons acting jointly or in concert, other than an agent, who makes an Offer;
- (e) "Offeror's Notice" means the notice described in Section 14.6(4); and
- (f) "Offeror's Units" means Units (including all Units issuable to the Offeror upon the exchange or conversion of all Exchangeable Securities held by or on behalf of the Offeror) beneficially owned, or over which control or direction is exercised, on the date of an Offer by the Offeror or any affiliate or associate of the Offeror;
- (2) In the event of an Offer, any holder of Exchangeable Securities may, unless otherwise prohibited by the terms and conditions of the Exchangeable Securities, convert, exercise or exchange (whether conditionally or otherwise), such Exchangeable Securities, as the case may be, into Units for the purpose of tendering the Units acquired on such conversion, exercise or exchange to such Offer. Such conversion, exercise or exchange, will be conditional upon the taking up by the Offeror of Units pursuant to the Offer.
- (3) If an Offer is made for all outstanding Units, by such Offer, the Offeror agrees to be bound by the provisions of this Article 14, and:
 - (a) within the time provided in the Offer for its acceptance or within 120 days after the date the Offer is made, whichever period is the longer, the Offer is accepted by the holders of at least 90% of the Units (on a fully diluted basis, assuming the exchange of all Exchangeable Securities for Units) other than the Offeror's Units; and
 - (b) the Offeror is bound to take up and pay for, or has taken up and paid for the Units of the holders of Units who accepted the Offer;

then:

- (c) if any holder of Exchangeable Securities has not either transferred such Exchangeable Securities to the Offeror or elected to convert (whether conditionally or otherwise) such Exchangeable Securities for Units, then the Offeror will be entitled to require the Trust to require the automatic exchange of Exchangeable Securities in accordance with the terms of any applicable Exchange Agreement; and
- (d) the Offeror will be entitled to acquire, and the Dissenting Offeree's will be required to sell to the Offeror, the Units held by the Dissenting Offeree's on the terms on

which the Offeror acquired the Units of the Unitholders who accepted the Offer, provided that the Offeror complies with Section 14.6(4) and 14.6(6).

- (4) Where an Offeror is entitled to require the Trust to require the automatic exchange of Exchangeable Securities and/or to acquire Units held by a Dissenting Offeree pursuant to Section 14.6(3), and the Offeror wishes to exercise such right, the Offeror shall send by registered mail, before the earlier of (i) 60 days after the date of termination of the Offer and (ii) 180 days following the date of Offer, notice (the "Offeror's Notice") to the Trust, each holder of Exchangeable Securities and/or to each Dissenting Offeree stating that:
 - (a) Holders holding at least 90% of the Units (on a fully diluted basis, assuming the exchange of all Exchangeable Securities), other than the Offeror's Units, have accepted the Offer;
 - (b) the Offeror has taken up and paid for the Units of the Unitholders who accepted the Offer;
 - (c) as applicable, each Exchangeable Security will be automatically converted or exchanged for Units;
 - (d) a Dissenting Offeree must transfer his, her or its respective Units to the Offeror on the terms on which the Offeror acquired the Units of the Unitholders who accepted the Offer within 21 days after the date of the sending of the Offeror's Notice; and
 - (e) Dissenting Offeree's must send the certificates representing their respective Units to the Trust, duly endorsed for transfer, within 21 days after the date of the sending of the Offeror's Notice.
- (5) A holder of Exchangeable Securities and/or a Dissenting Offeree to whom an Offeror's Notice is sent pursuant to Section 14.6(4), shall, within 21 days after the sending of the Offeror's Notice, as applicable, send the certificate representing his, her or its Units (or, as applicable, Exchangeable Securities) to the Trust, duly endorsed for transfer, if a certificate for such Units has been provided.
- (6) Within 21 days after the Offeror sends an Offeror's Notice pursuant to Section 14.6(4), the Offeror shall pay or transfer to the Trustees, or to such other person as the Trustees may direct, the cash or other consideration that the Offeror would have had to pay to the Dissenting Offeree had they accepted the Offer in respect of the Units that they continue to hold (including the Units issuable upon the conversion or exchange of any Exchangeable Securities held by the Dissenting Offeree).
- (7) The Trustees, or the person directed by the Trustees, shall hold in trust for the Dissenting Offeree's the cash or other consideration received under Section 14.6(6), but such cash or other consideration shall not form any part of the Trust Assets. The Trustees, or such persons, shall deposit such cash in a separate account in a Canadian chartered bank, and shall place such other consideration in the custody of a Canadian chartered bank or similar institution, for safekeeping.
- (8) Within 30 days after the date of the sending of an Offeror's Notice pursuant to Section 14.6(4), the Trustees, if the Offeror has complied with Section 14.6(6), shall:

- (a) do or cause to be done all acts and things and execute and cause to be executed all instruments as in the opinion of the Trustees may be necessary or desirable to effect the conversion or exchange of all Exchangeable Securities for Units and shall cause the transfer of the Units of the Dissenting Offeree's (including, as applicable, the Units issuable to such Dissenting Offeree's upon the conversion of their Exchangeable Securities) to the Offeror;
- (b) send or cause to be sent to each Dissenting Offeree who has complied with Section 14.6(5) above (as applicable) the consideration to which such Dissenting Offeree is entitled under this Section 14.6; and
- send or cause to be sent to each Dissenting Offeree who has not complied with Section 14.6(5) (as applicable) a notice stating that:
 - (i) as applicable, his, her or its Exchangeable Securities have been converted, exercised or exchanged for Units;
 - (ii) his, her or its Units have been transferred to the Offeror;
 - (iii) the Trustees or some other person designated in such notice are holding in trust the consideration for such Units; and
 - (iv) the Trustees, or such other person, will send the consideration to such Dissenting Offeree as soon as practicable after receiving the certificate(s) representing such Dissenting Offeree's Units (or, as applicable, Exchangeable Securities) or such other documents as the Trustees or such other person may require in lieu thereof,

and the Trustees are hereby appointed the agent and attorney of the Dissenting Offeree for the purposes of giving effect to the foregoing provisions.

- (9) Subject to applicable law, an Offeror cannot make an Offer for Units unless, concurrent with the communication of the Offer to any Unitholder, a copy of the Offer is provided to the Trust.
- (10) For the purpose of giving effect to the provisions hereof, the Trust shall cause the terms, conditions, restrictions, rights and obligations of Exchangeable Securities to contain the provisions necessary to give effect to this Section 14.6.

ARTICLE 15 TERMINATION

15.1 Term of Trust

Subject to the other provisions of this Declaration of Trust, the Trust shall continue for a term ending 21 years after the date of death of the last surviving issue of Her Majesty Queen Elizabeth II alive on March 27, 2007. For the purpose of terminating the Trust by such date, the Trustees will commence to wind-up the affairs of the Trust on such date as may be determined by the Trustees, being not more than two years prior to the end of the term of the Trust.

15.2 Extension or Termination with the Approval of Voting Unitholders

The Voting Unitholders may vote by Special Resolution to extend the term of or terminate the Trust at any meeting of Voting Unitholders duly called by the Trustees for the purpose of considering the extension or termination of the Trust. If Voting Unitholders vote to terminate the Trust, the Trustees will commence to wind-up the affairs of the Trust. Such Special Resolution may contain such directions to the Trustees as the Voting Unitholders determine, including a direction to distribute the Trust Assets, in specie, subject to compliance with any securities or other laws applicable to such distributions.

15.3 Procedure Upon Termination

Following the passage of a Special Resolution authorizing the termination of the Trust, or upon being required to commence the wind-up of the affairs of the Trust, the Trustees will give notice thereof to the Voting Unitholders, which notice will designate the time or times at which Voting Unitholders shall surrender certificates representing their Units and Special Voting Units for cancellation and the date on which the Register will be closed.

15.4 Powers of the Trustees Upon Termination

After the date on which the Trustees are required to commence winding-up the affairs of the Trust, the Trustees will undertake no activities with respect to the Trust except for the purpose of winding-up the affairs of the Trust and, for this purpose, the Trustees will continue to be vested with and may exercise all or any of the powers conferred upon the Trustees under this Declaration of Trust.

15.5 Sale of Investments

Following the closure of the Register, the Trustees will proceed to wind-up the affairs of the Trust as soon as may be reasonably practicable, and for such purpose may, subject to any direction to the contrary given in respect of a termination authorized under Section 15.2, cause the Trust to fulfill or discharge the contracts of the Trust, perform or cause the Auditor to perform any final audit of the Trust Assets, cause the Trust to collect its assets, sell, convey, assign, exchange, transfer or otherwise dispose of all or any part of the remaining Trust Assets, to one or more persons in one transaction or a series of transactions at public or private sale for consideration that may consist in whole or in part of cash, securities or other property of any kind, discharge or pay its liabilities, and do all other acts appropriate to liquidate the Trust, and shall in all respects act in accordance with the directions, if any, of the Voting Unitholders (given in respect of a termination authorized under Section 15.2), subject to compliance with applicable laws.

15.6 Distribution of Proceeds or Assets

After causing the Trust to pay, retire or discharge or make provision for the payment, retirement or discharge of all known liabilities and obligations of the Trust and to provide for indemnity against any other outstanding liabilities and obligations, the Trustees will, subject to obtaining all necessary regulatory approvals, distribute the remaining part of the proceeds of the sale of its assets among the Unitholders in accordance with their pro rata interests. If the Trust is unable to sell all or any of the assets of the Trust by the date set for termination, the Trustees may, subject to

obtaining all necessary regulatory approvals, distribute the remaining assets directly to the Unitholders in accordance with their pro rata interests. The Trustees will have no liability to the Trust or any Unitholder for any amounts received by the Trust or the Unitholders in connection with the liquidation of the Trust, provided that the Trustees act honestly and in good faith with a view to the best interests of the Trust.

15.7 Further Notice to Voting Unitholders

In the event that less than all of the Voting Unitholders have surrendered their certificate(s) representing their Voting Units for cancellation within six months after the time specified in the notice referred to in Section 15.3, the Trustees will give further notice to the remaining Voting Unitholders to surrender the certificate representing their Voting Units for cancellation and if, within one year after the further notice, all certificates representing the Voting Units have not been surrendered for cancellation, such remaining Voting Units will be deemed to be cancelled without prejudice to the rights of the holders of Voting Units comprising such Voting Units, if applicable, to receive their pro rata share of the remaining Trust Assets, and the Trustees may either take appropriate steps, or appoint an agent to take appropriate steps, to contact such Voting Unitholders (deducting all expenses thereby incurred from the amounts to which such Voting Unitholders are entitled as aforesaid, if applicable) or, in the discretion of the Trustees, may pay such amounts into court, and/or deposit such amounts in an account in a chartered bank or similar institution in Canada in the name of such Voting Unitholder, if applicable, for delivery against receipt of Voting Units for cancellation, and the Trust, the Trustees and any representative thereof shall thereupon be released from any and all further liability with respect to such property and thereafter the Voting Unitholder will have no rights as against the Trust, the Trustees or any representative thereof in respect of such property or an accounting therefor.

15.8 Responsibility of the Trustees after Sale and Conversion

The Trustees will be under no obligation to invest the proceeds of any sale of investments or other assets or cash forming part of the Trust Assets after the date referred to in Section 15.3 and, after such sale, the sole obligation of the Trustees under this Declaration of Trust will be to hold such proceeds or assets in trust for distribution under Section 15.6 or Section 15.7.

ARTICLE 16 SUPPLEMENTAL INDENTURES

16.1 Provision for Supplemental Indentures for Certain Purposes

The Trustees may, without approval of the Unitholders and subject to the provisions hereof, and shall, when so required pursuant to the provisions hereof, execute and deliver indentures or instruments supplemental hereto which thereafter shall form part hereof for any one or more or all of the following purposes:

(a) amending any provisions of this Declaration of Trust in the circumstances set forth in Section 11.1 where the Trustees may do so without the consent, approval or ratification of the Unitholders or any other person; and

(b) amending any provisions of this Declaration of Trust where the modification or amendment has been approved by an Ordinary Resolution or by Special Resolution, as the case may be.

ARTICLE 17 GENERAL

17.1 Notices

- (1) Any notice or other document required to be given or sent to Voting Unitholders under this Declaration of Trust and any and all other communications to Voting Unitholders will be deemed to have been duly given if sent through ordinary post addressed to each Voting Unitholder at his or her address of record on the Register or by such other method designed to give reasonable general notice thereof as is determined by the Trustees, which may include publication in a newspaper having national circulation in Canada. Any notice so given will be deemed to have been given on the day that is five days following that on which the notice or other communication was posted or, in the case of notice being given by another method determined by the Trustees, on the day specified by the Trustees. In proving notice was posted, it will be sufficient to prove that such letter or circular was properly addressed, stamped and posted.
- (2) Any notice, direction or other communication to be given or sent to the Trust shall be in writing and given by delivering, mailing or sending it by facsimile to the head office of the Trust or in such other manner as the Trustees may direct. Any such communication will be deemed to have been validly and effectively given (i) if personally delivered, on the date of delivery, (ii) if mailed by registered mail, five days from the date of mailing, (iii) if transmitted by facsimile, on the Business Day following the date of confirmation of transmission, and (iv) if sent by some other manner as directed by the Trustees, upon receipt by the Trust. If any such notice or communication has been mailed and if regular mail service is interrupted by strikes or other irregularities, such notice or communication will be deemed to have been received 48 hours after 12:01 a.m. on the day following the resumption of normal mail service.

17.2 Failure to Give Notice

The failure by the Trustees, by accident or omission or otherwise unintentionally, to give any Voting Unitholder any notice provided for herein will not affect the validity, effect or taking effect of any action referred to in such notice, and the Trustees will not be liable to any Voting Unitholder for any such failure.

17.3 Joint Holders

Service of a notice or document on any one of several joint holders of Voting Units will be deemed effective service on the other joint holders.

17.4 Service of Notice

Any notice or document sent by post to or left at the address of a Voting Unitholder pursuant to this Article will, notwithstanding the death, disability or bankruptcy of such Voting Unitholder, and whether or not the Trustees have notice of such death, disability or bankruptcy, be deemed to

have been fully served and such service will be deemed sufficient service on all persons having an interest in the Voting Units concerned.

17.5 Information Available to Voting Unitholders

Each Voting Unitholder has the right to obtain, on demand and without fee, from the head office of the Trust a copy of this Declaration of Trust and any amendments thereto, and will be entitled to examine a list of Voting Unitholders.

17.6 Fiscal Year

The fiscal year of the Trust will end on December 31 of each year.

17.7 Financial Disclosure

The Trust will send (or otherwise make available electronically on such website of the Trust as the Trust or its affiliates may maintain if sending is not required by law) to Voting Unitholders such consolidated financial statements of the Trust and such other reports as are from time to time required by law, including:

- (a) at least 21 days prior to the date of each annual meeting of Voting Unitholders (or such other time period as may be required by applicable law), the annual financial statements of the Trust for the fiscal year ended immediately prior to such annual meeting, together with comparative financial statements for the preceding fiscal year, if any, and the report of the Auditors thereon referred to in Section 18.4; and
- (b) within 45 days (or such shorter time period as may be required by applicable law) after the end of each fiscal quarter of the Trust (other than the fourth quarter of each year), unaudited quarterly financial statements of the Trust for such fiscal quarter, together with comparative financial statements for the same fiscal quarter in the preceding fiscal year, if any.

Such financial statements will be prepared in accordance with GAAP; provided that such statements and the obligations to deliver such statements may vary from such principles to the extent required to comply with applicable securities laws or securities regulatory requirements or to the extent permitted by applicable securities regulatory authorities.

17.8 Voting Unitholder Meeting Information

Prior to each meeting of Voting Unitholders, the Trust will provide to each Voting Unitholder, together with the notice of the meeting:

- (a) a form of proxy which can be used by a Voting Unitholder to appoint a proxy, who need not be a Voting Unitholder, to attend and act at the meeting on behalf of the Voting Unitholder, in the manner and to the extent authorized by the proxy; and
- (b) all information required by applicable law.

17.9 Taxation Information

On or before March 15 in each year, the Trust will provide to Unitholders who received distributions from the Trust in the prior calendar year such information regarding the Trust as is required by Canadian federal and provincial law to be submitted to Unitholders for Canadian income tax purposes to enable Unitholders to complete their Canadian tax returns in respect of the prior calendar year.

17.10 Delegation of Trust Management

The Trustees shall be authorized from time to time to engage independent contractors to manage the Trust and to delegate to such contractors such duties as the Trustees consider appropriate. Without limiting the foregoing, the Trust shall be authorized to enter into the Management Agreement with the Manager providing for certain officers and employees of the Manager to provide management services to the Trust and providing for the supply to the Trust of various management, administrative and support services, including accounting and human resource services, office space and equipment and the necessary clerical and secretarial personnel for the administration of the day-to-day activities of the Trust. The Trustees shall be authorized to execute the Management Agreement and settle the terms and conditions thereof, and any amendments thereto, and to terminate the Management Agreement in accordance with its terms.

17.11 Power of Attorney

- (1) The Trustees hereby grant to the Manager, a power of attorney constituting the Manager, with full power of substitution, as their true and lawful attorney to act on behalf of the Trust with full power and authority in their name, place and stead, and to execute, under seal or otherwise, swear to, acknowledge, deliver, make or file or record when, as and where required, any instrument, deed, agreement or document in connection with carrying out the activities of the Trust including, without limitation, to execute and deliver, on the Trust's behalf, any Exchange Agreement, the Registrar and Transfer Agency Agreement and such other agreements, documents, certificates and other writings as are required or contemplated by the Information Circular, the Closing or this Declaration of Trust.
- (2) Each Voting Unitholder hereby grants to the Trustees and their successors and assigns, a power of attorney constituting the Trustees and such successors and assigns with full power of substitution, as his or her true and lawful attorney to act on his or her behalf, with full power and authority in his or her name, place and stead, and to execute, under seal or otherwise, swear to, acknowledge, deliver, make or file or record when, as and where required:
 - (a) this Declaration of Trust and any other instrument required or desirable to qualify, continue and keep in good standing the Trust as a mutual fund trust;
 - (b) any instrument, deed, agreement or document in connection with carrying on the affairs of the Trust as authorized in this Declaration of Trust including all conveyances, transfers, and other documents required in connection with any disposition of securities required under Sections 5.7 and 14.1;

- (c) all conveyances, transfers and other documents required in connection with dissolution, liquidation or termination of the Trust in accordance with the terms of this Declaration of Trust;
- (d) any and all elections, determination or designations whether jointly with third parties or otherwise, under the Tax Act or any other taxation or other legislation or similar laws of Canada or of any jurisdiction in respect of the affairs of the Trust or of a Voting Unitholder's interest in the Trust; and
- (e) all transfer, conveyances and other documents required to facilitate the acquisition of Voting Units, Exchangeable Securities of non-tendering offerees pursuant to Section 14.6.

The power of attorney granted herein is, to the extent permitted by applicable law, a power coupled with an interest and is irrevocable and will survive the assignment by each Voting Unitholder of all or part of the Voting Unitholder's interest in the Trust and will extend to and bind the heirs, executors, administrators and other legal representatives and successors and assigns of the Voting Unitholder.

17.12 No Breach

Notwithstanding any other provision of this Declaration of Trust, Voting Unitholders shall have no power to effect any amendment hereto which would require the Trustees to take any action or conduct the affairs of the Trust in a manner which would constitute a breach or default by the Trust or the Trustees under any agreement binding on or obligation of the Trust or the Trustees.

17.13 Execution of Instruments

Any two Trustees and/or officers of the Trust shall have the authority to sign in the name of and on behalf of the Trust all instruments in writing and any instruments in writing so signed shall be binding upon the Trust without any further authorization or formality. For greater certainty, the foregoing shall not limit in any way the power of the Trustees to delegate to any person or persons, including without limitation the Manager, the authority to sign in the name of and on behalf of the Trust any specific instrument or any instruments in writing generally.

17.14 Trust Property

The Trustees shall maintain the Trust's property separate from all other property in their possession and from the property of all other persons.

17.15 Trust Records

The Trustee shall prepare and maintain, at the principal office of the Trust or at any other place in Canada designated by the Trustees, records containing: (i) the Declaration of Trust; and (ii) minutes of meetings and resolutions of the Unitholders. The Trust shall also prepare and maintain adequate accounting records and records containing minutes of meetings and resolutions of the Trustees and any committee thereof. Such records shall be kept at the principal office of the Trust or at such other place in Canada as the Trustees think fit and shall at all reasonable times be open to inspection by the Trustees.

ARTICLE 18 AUDITORS

18.1 Qualification of Auditors

The Auditors shall be an independent recognized firm of chartered accountants which has an office in Canada.

18.2 Appointment of Auditors

The initial Auditors, which shall hold office until the first annual meeting of the Voting Unitholders, will be appointed by a written resolution of the Initial Unitholder, as sole Unitholder, prior to the Closing. Subject to Section 18.3, the Auditors will be selected at each succeeding annual meeting of Unitholders. The Auditors will receive such remuneration as may be approved by the Trustees. If at any time a vacancy occurs in the position of Auditors, the Trustees may appoint new auditors to act as the Auditors until the next annual meeting of the Unitholders.

18.3 Change of Auditors

The Auditors may at any time be removed by the Trustees with the approval of the Voting Unitholders by Ordinary Resolution passed at a meeting of Unitholders duly called for the purpose and, upon the resignation or the removal of Auditors as aforesaid, new auditors may be appointed by Ordinary Resolution passed at a meeting duly called for the purpose or, in the absence of such meeting, by the Trustees.

18.4 Report of Auditors

The Auditors shall audit the accounts of the Trust at least once in each year and a report of the Auditors with respect to the annual financial statements of the Trust shall be provided to each Voting Unitholder with the annual financial statements referred to in Section 17.7.

ARTICLE 19 MISCELLANEOUS

19.1 Counterparts

This Declaration of Trust may be executed in several counterparts, each of which when executed shall be deemed to be an original, and such counterparts, together, shall constitute but one and the same instrument, which shall be sufficiently evidenced by any such original or facsimile counterparts.

19.2 Severability

The provisions of this Declaration of Trust are severable. If any provision of this Declaration of Trust shall be held invalid or unenforceable in any jurisdiction, such invalidity or unenforceability shall attach only to such provision in such jurisdiction and shall not in any manner affect or render invalid or unenforceable such provision in any other jurisdiction or any other provision of this Declaration of Trust in any jurisdiction.

19.3 Successors and Assigns

The provisions of this Declaration of Trust shall enure to the benefit of, and be binding upon, the parties hereto and their respective heirs, executors, administrators, successors and assigns.

19.4 References to Agreements

Any reference herein to any agreement, contract, indenture or obligation shall refer to such agreement, contract, indenture or obligation as the same may be amended from time to time.

19.5 Language

Les parties aux presentes ont exiges que la presente convention ainsi que tons les documents et avis qui s'y rattachent et/ou qui en decouleront soient rediges en la langue anglaise. The parties hereto have required that this Declaration of Trust and all documents and notices resulting herefrom be drawn up in English.

[Remainder of this page intentionally left blank.]

IN WITNESS WHEREOF this Declaration of Trust has been executed by the undersigned as of the date referenced above.

"Witness"	"Paul Dykeman"
Witness	Paul Dykeman, Trustee
"Witness"	"Adam Gant"
Witness	Adam Gant, Trustee
"Witness"	"John van Haastrecht"
Witness	John van Haastrecht, Trustee
"Witness"	"Louis Maroun"
Witness	Louis Maroun, Trustee
"Witness"	"Patrick Miniutti"
Witness	Patrick Miniutti, Trustee
"Witness"	"Saul Shulman"
Witness	Saul Shulman, Trustee